Integrating Circular Economy Principles into Systems Engineering Processes for the Construction Industry

MSc Thesis

Department of Construction Management & Engineering

Abdulrahman Ali Mohammed Al-Yamani

August, 2024



UNIVERSITY OF TWENTE.

Preface

I am proud to share with you my master's thesis, the final achievement of my journey as a master's student in the Construction Management and Engineering department at the University of Twente. This research, which uniquely explores the integration of Circular Economy principles into the Systems Engineering processes for the construction industry, reflects my belief in the critical importance of innovative practices for the future of construction. It is a culmination of my desire to learn, grow, and contribute meaningfully to the future of the construction industry.

This research was conducted in collaboration with ProcessMinded, which was an enriching experience, and provided the ideal environment for carrying out this research. Their innovative approach and expertise in Systems Engineering allowed me to delve deeply into the subject's complexities. I am profoundly grateful to my supervisors and teachers from the University of Twente, Robin de Graaf and Marc van den Berg, for their invaluable guidance and support. I also sincerely thank Wilco van Roekel and Ruben Herrebrugh, my supervisors from ProcessMinded, whose practical insights were instrumental throughout my research.

I am deeply grateful to my family and friends, whose support has been the bedrock of my academic journey. Their love and encouragement have been my guiding light, helping me navigate through the toughest moments. In particular, I want to express my heartfelt thanks to my mother, whose unwavering belief in me has been a constant source of strength. As Nietzsche wisely said, "He who has a why can bear any how" and it is your support that has helped me bear every 'how' along the way.

Lastly, I hope this thesis contributes to the ongoing dialogue in improving construction practices and encourages others to push the boundaries of what is possible in our field.

Enjoy the read.

Abdulrahman Alyamani Enschede, August 2024

Colophon

Title:

Integrating Circular Economy Principles into Systems Engineering Processes for the Construction Industry

Written by:

Abdulrahman Ali Mohammed Al-Yamani Department of Construction Management & Engineering, University of Twente

Supervised by:

dr. ir. Robin S. de Graaf^a dr. ir. Marc van den Berg^a ir. Wilco van Roekel^b ir. Ruben Herrebrugh^b

Affiliations:

^a – University of Twente
 https://www.utwente.nl
 ^b – ProcessMinded
 https://www.processminded.nl/

Copyright © 2024 by A.A.M Al-Yamani

All rights reserved. No part of this publication may be reproduced, distributed, or transmitted in any form or by any means, including photocopying, recording, or other electronic or mechanical methods, without the prior written permission of the author.

Contact Information: Email: a.a.m.alyamani1@gmail.com

UNIVERSITY OF TWENTE.



Integrating Circular Economy Principles into the Systems Engineering Process for the Construction Industry

Abdulrahman Ali Mohammed Al-Yamani

August, 2024

Abstract

The construction industry faces significant challenges in resource depletion and waste generation, necessitating a shift towards more circular and sustainable practices. This study, conducted in collaboration with ProcessMinded, a consultancy firm specializing in Systems Engineering (SE) in construction projects, addresses these challenges by investigating the integration of circular economy principles into the SE process. Focusing on four key circularity principles: closing, narrowing, slowing, and regenerating the resource loops. The research develops a framework to guide the implementation of these principles across each step of the SE process: Project start, requirements analysis, functional analysis and allocation, design synthesis, and verification and validation. The methodology comprised a narrative literature review synthesizing diverse sources for the integration. This literature review informed the development of a structured framework organizing circularity integration for each SE step into inputs, actions, and outputs. The framework also includes practical examples for some actions to illustrate real-world applications. To validate the framework, semi-structured interviews were conducted with five experienced systems engineers from ProcessMinded. These interviews provided insights into the framework's strengths, potential improvements, and applicability of implementing the suggested actions within the framework in practice. The project start and verification steps showed particular promise for circularity integration, according to interviewees, while the design synthesis step emerged as more challenging. The resulting framework offers a systematic approach to integrating circular economy principles within the SE process, from project start through to verification and validation. It addresses the need for sector-specific, interdisciplinary approaches to circular economy implementation in construction, as identified in recent literature. This study contributes both to the theoretical understanding of circularity integration in SE and practical applications within the construction industry. It provides a foundation for further research and implementation of circular economy principles in construction projects through the lens of SE. Future research should focus on refining the application of the framework in diverse contexts, considering the iterative nature of SE and the varying levels of detail in the SE process and possibly addressing the challenges identified across different SE steps.

Contents

1	INT	RODU	JCTION	1			
	1.1	Backg	round	1			
	1.2	Resear	ch Focus	2			
	1.3	Resear	ch Questions	4			
	1.4	Struct	ure of the Report	4			
2	Met	thodol)gv	5			
	2.1	Resear	ch Design	$\overline{5}$			
	2.2	Literat	ture Review (Step 1-3)	5			
		2.2.1	Search Strategy	6			
		2.2.2	Step 1: Identifying Processes and Principles	6			
		223	Step 2: Linking SE processes and CE principles	7			
		2.2.0 2.2.4	Step 3: Building Framework	$\frac{1}{7}$			
	2.3	Valida	tion Interviews (Step 4-5) \ldots \ldots \ldots \ldots \ldots \ldots	7			
9	C	toma T	Angineering in Construction	0			
3	Sys	tems E	angineering in Construction	9			
	3.1	Systen	ns Engineering Across industries	9			
	3.2	Systen	Is Engineering Process and the Vee Model	0			
		3.2.1	Process input (Project Start) II	0			
		3.2.2	Requirements Analysis	1			
		3.2.3	Functional Analysis and Allocation	1			
		3.2.4	Design Synthesis	2			
		3.2.5	Verification and Validation	2			
4	Cire	Circular Economy Principles in Construction 13					
	4.1	Frame	works and Principles of Circular Economy in Construction 13	3			
	4.2	The F	our Circular Economy Principles	4			
		4.2.1	Closing the Resource Loop (Use Again)	4			
		4.2.2	Narrowing the Resource Loop (Use Less)	5			
		4.2.3	Slowing the Resource Loop (Use Longer)	5			
		4.2.4	Regenerating the Resource Loop (Make Clean)	5			
5	Cir	cular S	vstems Engineering Process 10	6			
Ŭ	5.1	Projec	t Start	7			
	5.2	Requi	rements Analysis	8			
	5.3	Functi	onal Analysis and Allocation	0			
	5.4	Design	Synthesis	1			
	55	Vorifie	ation and Validation	л Т			
	5.6	Framo	work and Examples 2	5			
	5.0	561	Circularity Integrated SE Process Framework	5			
		5.6.2	Examples from Practice	8			
~	F			~			
6	Fra	mewor.	k Validation 30	U			
	6.1	Procee	iure and Interviewee Details	U			
	6.2	Valida	tion Insights Per SE Step	0			
		6.2.1	Circularity Integrated Project Start	U			

		6.2.2	Circularity Integrated Requirements Analysis	31	
		6.2.3	Circularity Integrated Functional Analysis and Allocation	31	
		6.2.4	Circularity Integrated Design Synthesis	32	
		6.2.5	Circularity Integrated Verification and Validation	32	
		6.2.6	Framework Validation Summary and Conclusion	33	
7	7 Discussion				
	7.1	Summ	nary of Key Findings	34	
	7.2	Resea	rch Contribution	35	
	7.3	Limita	ations and Future Research	36	
		7.3.1	Limitations	36	
		7.3.2	Future Research	37	
8	Cor	nclusio	n	39	

1 INTRODUCTION

1.1 Background

The construction industry faces challenges as projects grow increasingly complex and global sustainability concerns intensify. These challenges arise in construction projects due to multiple factors: the involvement of multiple disciplines, a high number of stakeholders, large-scale operations, extended lifespans and execution periods, intricate decision-making processes, significant environmental impacts, and associated risks (de Bruijn & ten Heuvelhof, 2018; Maylor & Turner, 2017; Qazi et al., 2016; Shen et al., 2010; Wood & Ashton, 2010). Simultaneously, the world struggles with resource depletion, excessive waste generation, and escalating environmental damage. The construction industry is a major consumer of resources and generator of waste, accounting for approximately 32% of the world's resources (Yeheyis et al., 2013), and finds itself at the center of these challenges, necessitating a paradigm shift in its practices.

In response to these challenges, two distinct but potentially complementary approaches have gained attention: Systems Engineering (SE) and Circular Economy (CE) principles. SE, with its origins in post-World War II mainly from defense and aerospace industries, offers a holistic methodology for managing complex projects (Ferris, 2007a, 2007b; N. U. I. Hossain et al., 2020; Reid & Wood, 2023). It emphasizes cross-disciplinary and transdisciplinary integration, continuous evolution, and adaptation to complex technical systems (Brook et al., 2024; Dykes et al., 2011; Mesmer et al., 2022; M. Watson et al., 2022; M. D. Watson et al., 2016). SE, which is seen as a comprehensive approach to managing complex systems, has gained notable momentum in the construction industry due to its ability to handle the challenges of complex construction projects (O. Hoehne, 2023; O. M. Hoehne, 2012). Furthermore, SE is characterized by its comprehensive approach to project management and system design. According to the International Council on System Engineering (INCOSE), SE is defined as:

An interdisciplinary approach and means to enable the realization of successful systems. It focuses on defining customer needs and required functionality early in the development cycle, documenting requirements, and proceeding with design synthesis and system validation while considering the complete problem: operations, cost and schedule, performance, training and support, test, manufacturing, and disposal. Systems Engineering integrates all the disciplines and specialty groups into a team effort, forming a structured development process that proceeds from concept to production to operation. Systems Engineering considers both the business and the technical needs of all customers with the goal of providing a quality product that meets the user needs. (INCOSE, 2015, p. 11)

This comprehensive definition highlights SE's potential to address the complexities inherent in construction projects. In the Netherlands, major clients like Rijkswaterstaat, the executive agency of the Ministry of Infrastructure and Water Management, and Prorail, the governmental organization responsible for the national railway network, have mandated SE application in their projects since 2007 (Leidraad SE., 2007). Concurrently, the circular economy model has emerged as a potential solution to the construction industry's sustainability challenges. The circular economy concept, rooted in Kenneth Boulding's 1960s "Closed" or "Spaceman Economy" idea (Boulding, 1966), has been further developed by organizations like the Ellen MacArthur Foundation (Ellen MacArthur Foundation, 2013), encouraging the construction industry to reconsider material use and waste management approaches. Circular economy principles aim to transition the industry from a linear "take-make-dispose" model to a more sustainable, circular approach. This shift is crucial, given that the construction industry generates about 37.5% of Europe's total waste (Eurostat, 2023). The urgency of adopting circular economy principles in the construction industry is emphasized by ambitious sustainability goals, such as the Netherlands aim to reduce emissions by 80% to 95% compared to 1990 levels by 2050 (Government of the Netherlands, 2019). This objective must be balanced with the country's plans to construct 900,000 new houses by 2030 (International Monetary Fund. European Dept., 2023), with the required infrastructure development with such a goal, highlighting the need for sustainable practices in an expanding industry.

The complexity and breadth of circular economy concepts have led to an increased number of definitions in the literature, with varying perceptions of these concepts. For example, Kirchherr et al. (2017) analyzed 114 definitions of circular economy to come up with the 115th definition. However, this research focuses on four comprehensive circular economy principles based on the work of N. Bocken et al. (2021), N. M. Bocken et al. (2016), and Çetin et al. (2021). These four principles are particularly relevant to the construction industry and are as follows: **Closing** (reusing and recycling materials to close the loop), **Narrowing** (using fewer resources per product), **Slowing** (extending the lifespan of products), and **Regenerating** the resource loop (using renewable materials and energy). These circular economy principles could offer a promising approach to addressing circularity in construction. Nonetheless, their implementation requires integration with existing practices like SE, which forms the core of this research. The focus of this research is elaborated on in the following subsection.

1.2 Research Focus

This research focuses on integrating circular economy principles into the SE process within the context of the construction industry. The research is motivated by and conducted in close collaboration with ProcessMinded, a consultancy firm that applies SE methodologies to a wide range of construction projects for various clients and has been operating since 2013. ProcessMinded acknowledges the challenges and the necessity for change in current practices to align with national and international sustainability goals. The company is actively seeking to further enhance its application of SE and aims to explore a widely applicable approach to incorporate circularity into SE. Due to the variety of projects they work on, their interest is to adopt a circular approach for SE that is generic enough to be used across different projects yet flexible enough to incorporate different circular economy principles

The integration of SE and circular economy remains relatively unexplored, particularly in practical and industry-specific applications within the construction industry. This research aims to address this gap by investigating the integration of circular economy principles into SE processes for construction projects. Previous attempts to integrate circular economy principles within the construction industry have utilized various management approaches, including Construction and Demolition Waste (CDW) management (Gálvez-Martos et al., 2018), systemic thinking, and multi-level circular economy methods (Sparrevik et al., 2021; Többen & Opdenakker, 2022). The ReSOLVE framework (Regenerate, Share, Optimize, Loop, Virtualise, and Exchange) has also been applied to enhance circularity in the construction industry and integrate circularity at early project stages (Iyer-Raniga, 2019). Additionally, a Circular Economy Interface Matrix Analysis (CEIMA) framework has been developed to identify critical interfaces between the circular economy and stakeholders, offering practical guidance for infrastructure organizations to transition toward circular practices (Coenen et al., 2020). However, these approaches have not explicitly addressed integration with SE methodology in the context of the construction industry.

This research strives to leverage these existing attempts while analyzing both SE and circular economy paradigms separately to identify possible concept synergies and potential integration points. For instance, established SE practices for defining performance targets could be enhanced to address the lack of knowledge about the construction industry's environmental performance, which has hindered broader circularity adoption (Eberhardt et al., 2022). Additionally, circular economy principles could potentially mitigate uncertainty in early-stage design decisions within SE processes by providing guiding principles for circular goals (Van Der Meer et al., 2015). By combining the systematic approach of SE with a focus on circular economy principles, this research seeks to develop a framework that can guide ProcessMinded towards more circular applications of SE in their projects. In addition, this research contributes to the broader body of knowledge on SE and circular economy integration by responding to several critical needs identified in recent literature:

- 1. The call for more practical applications and frameworks for implementing circular economy principles in the construction industry (Antwi-Afari et al., 2021; Guerra et al., 2021; M. U. Hossain et al., 2020).
- 2. The need for sector-specific frameworks and interdisciplinary approaches identified by Gasparri et al. (2023) in their analysis of knowledge gaps for circular economy implementation in construction.
- 3. The demand for new design typologies and decision-making guides for circular economy implementation, as highlighted by Eberhardt et al. (2022).

Focusing on the integration of circular economy principles into SE, with Process-Minded as a key industry partner, this research aims to contribute to the body of knowledge in both SE and circular economy fields. The findings could have far-reaching implications, not only for ProcessMinded but also for other consultancies and organizations in the construction industry with similar challenges regarding circularity.

1.3 Research Questions

Building upon the identified need to integrate circular economy principles into the SE processes within the context of the construction industry and in collaboration with ProcessMinded, the main research question of this research is:

How can circular economy principles be integrated into systems engineering process, and what framework can be developed for ProcessMinded to implement this integration within the context of the construction industry?

To answer the main research question, this study sought to answer the following subquestions throughout the different phases of the research:

- **Q1:** What are systems engineering processes and circular economy principles in the construction industry?
- **Q2:** How can circular economy principles be integrated into systems engineering process?
- **Q3:** What framework can ProcessMinded use to integrate circular economy principles into systems engineering process?

These research questions align closely with the challenges and opportunities identified in the construction industry and in the context of ProcessMinded's work. By addressing these questions, this research seeks to contribute to the academic understanding of SE and circularity integration and practical applications for ProcessMinded within the construction industry.

1.4 Structure of the Report

This report explores the integration of circular economy principles into the systems engineering process within the context of the construction industry. It is important to note that the terms 'circular economy principles' and 'circularity principles' are used interchangeably throughout the report. The report is structured as follows: this section introduced the background of the research and presented the research focus and questions. Section 2 elaborates on the research methodology, detailing the research design, literature review process, and validation interviews. Section 3 provides an in-depth look at systems engineering in construction, explaining its application across industries and detailing the steps of the SE process adopted in this research. Section 4 explores circular economy principles in construction, discussing various circularity frameworks and explaining the four key principles adopted in this study. Section 5 presents the core of the research, integrating circular economy principles into each step of the SE process and developing a framework for implementation with supporting examples from practice. Section 6 covers the validation of the framework through expert interviews, presenting insights for each SE step. Section 7 discusses the key findings, research contributions, limitations, and future research directions. Finally, Section 8 concludes the report, summarizing the main outcomes.

2 Methodology

This section explains the research methodology, starting by presenting the research design and the research method. It then details how the research was conducted and how the three research questions were answered and ends with an explanation of the validation process conducted in this study.

2.1 Research Design

The goal of this research is to integrate circular economy principles into SE. To achieve this, a set of research questions was developed to guide the development of a framework for the integration aimed at SE practitioners in the construction industry. This research addresses the relatively unexplored intersection of circular economy principles and SE to offer a new theoretical perspective through a literature review and semi-structured interviews for validation. The sequence of activities and the research design to address each research question are explained in this section and illustrated in Figure 1 below.



Figure 1: Research Design

2.2 Literature Review (Step 1-3)

In this study, a narrative literature review was employed, as it allows for the discussion of topics from theoretical and contextual perspectives (Baumeister & Leary, 1997; Snyder, 2019). Narrative reviews use less formal methods, avoiding rigorous reporting and criteria and allowing for interpretation and critique (Ferrari, 2015; Greenhalgh et al., 2018). This approach was chosen for its flexibility in synthesizing diverse sources and its suitability for exploring the emerging topic of integrating circularity principles into systems

engineering processes. In addition, narrative reviews provide flexibility in searching for relevant literature and addressing the research questions, as they are suited for studies that do not aim to answer specific quantitative questions. Also, given the evolving nature of the circular economy and SE practices, no date restrictions were applied, providing a review of both historical and recent developments. Besides, the snowball technique was employed by reviewing references within identified sources that were relevant to the research topic. This technique helped uncover further studies that might not have been captured in the initial search (Badampudi et al., 2015).

2.2.1 Search Strategy

The literature search initially focused on the intersection of circular economy, systems engineering, and the construction industry using Scopus and Web of Science databases. The following query was used in the initial search attempt:

("circular economy" OR "circularity") AND ("systems engineering" OR "project management" OR "Reliability" AND "Availability" AND "Maintainability" AND "safety" OR "engineering management" OR "system thinking") AND ("construction industry" OR "built environment" OR "civil engineering")

After an initial screening of 52 unique records, it became evident that a different approach was needed to fully address the research questions. This is due to the fact that the results of this search were fragmented and not focused enough to capture the context of this research. Therefore, the search strategy was subsequently refined to better align with the research questions. The search involved linking specific SE processes with circularity principles, which guided a more focused review. This refined approach involved searching across multiple databases for themes connecting individual SE processes with circularity principles. Relevant grey literature was also included to provide a comprehensive view of current practices. The search strategy was tailored for each SE process, allowing for a detailed analysis and exploration of how circularity principles could be integrated into specific steps of systems engineering.

2.2.2 Step 1: Identifying Processes and Principles

This step involved gathering and synthesizing the theoretical knowledge related to SE, followed by the circular economy principles. Each was examined separately before integration. This approach helped build a foundational understanding that would inform the subsequent steps. This involved reviewing existing literature to identify the key processes of systems engineering for this research, as well as principles of circular economy within the context of the construction industry. SE principles were reviewed as well to comprehend the underlying reason for the technical processes commonly used in the construction industry, which have been addressed in relevant research and are recognized by authoritative organizations in the field of SE, such as the U.S. Department of Defense and INCOSE. Additionally, four circularity principles were identified and compared to existing frameworks, thereby addressing the research question (Q1).

2.2.3 Step 2: Linking SE processes and CE principles

The analysis process involved extracting key themes and patterns related to current practices and integration methods. By examining each SE process individually in relation to the four identified circular economy principles, the literature was synthesized to identify potential integration strategies. The analysis focused on highlighting how each SE process can incorporate the four circular economy principles and suggested potential actions for incorporating these principles. In addition, the analysis addressed potential challenges resulting from the integration and outlined ways to manage them. This has led to the second research question (Q2) being addressed.

2.2.4 Step 3: Building Framework

For each SE process analyzed, inputs, actions, and outputs for integrating circular economy principles were summarized. This systematic approach led to the development of a framework that included the intersection of the five SE processes with the four circularity principles, providing a practical guide for integrating circularity principles into the SE process for ProcessMinded. The focus on inputs, actions, and outputs was chosen as it is a well-established SE practice, as outlined in ISO/IEC/IEEE 15288 (2015) and the US Department of Defense (ISO/IEC/IEEE 15288, 2015; US Department of Defense, 2001), where every process is defined by what goes in, what happens, and what comes out. In addition, practical examples were provided to support and further explain how some of the actions presented in the framework can be done in practice. Lastly, this approach has helped in overcoming the complexity of the integration and simplifying the framework, which has led to answering the third research question (Q3)

2.3 Validation Interviews (Step 4-5)

To validate the proposed framework, semi-structured interviews were conducted with five professionals from ProcsessMinded who are experienced in SE. Semi-structured interviews are widely used in qualitative research and include a set of open-ended questions with the possibility to address questions that might emerge through the interview (DiCicco-Bloom & Crabtree, 2006). The five systems engineers have experience ranging from 8 to 17 years in the construction industry. In addition, two out of the five participants are currently in sustainability-related roles. Participants were chosen based on their experience in the field and familiarity with circularity concepts in engineering projects. An interview guide was developed with three open-ended questions and one quantitative question, covering each step of the framework: Project Start, Requirements Analysis, Functional Analysis and Allocation, Design Synthesis, and Verification and Validation. For each step of the SE process, participants were asked about:

- 1. What strengths (tops) do you see in these actions?
- 2. Do you have any tips to improve these actions for better integration with the systems engineering process?
- 3. Can you provide examples from your experience related to these actions?
- 4. On a scale from 1 to 5, with 5 being highly applicable and 1 being not applicable at all, how applicable do you think these actions are for systems engineers to implement? Why?

Actions were the primary focus during the validation process because they provided new insights to systems engineers. By focusing on the actions, the validation aimed to assess their relevance, validity, and applicability, ensuring that these actions could be seamlessly integrated into the conventional SE processes familiar to systems engineers. This focus allowed for a more practical evaluation of how circular economy principles could be incorporated within established SE practices. Furthermore, interviews were conducted online via Microsoft Teams via video conference. Each interview lasted approximately 45-65 minutes and covered all steps of the proposed framework. With participants' consent, the interviews were recorded and auto-transcribed by Microsoft Teams for analysis. Following the interviews, the transcripts were carefully reviewed and corrected by the researcher, and key insights, strengths, and areas for improvement were extracted for each step of the framework. Applicability ratings for the actions per step of SE in the framework were compiled and averaged across all participants. Noteworthy quotes illustrating important points or examples were identified for inclusion in the validation results presentation. The analysis focused on the direct interpretation of the interviewees' responses, identifying both common perspectives and insights relevant to each framework step.

3 Systems Engineering in Construction

This section discusses the application of SE within the construction industry, exploring its methodologies and processes. The discussion highlights how SE is tailored to meet industry-specific needs. The analysis follows the structure of SE as outlined by key industry standards and explains the steps of the SE process, with a focus on its application in construction projects.

3.1 Systems Engineering Across Industries

Although widely applied today, Systems Engineering can be understood in various ways depending on the systems under development, organizations, and industries considered. The fundamental principles of SE are lifecycle focus, managing interactions, ensuring compliance, and meeting stakeholder requirements. It is a dynamic field that continuously evolves, balancing cost, quality, and constraints while making informed decisions under uncertainty (Presland et al., 2018; M. Watson et al., 2022). Given the diverse applications of SE across industries, it is important to consider how different sectors adapt the method to their specific needs. SE is usually tailored to the specific systems and their environments, with methods changing based on the needs, operational context, and goals. According to M. D. Watson et al. (2016), evidence shows that 72% of surveyed companies employ unique systems engineering processes suited to their products. In the construction industry, Systems Engineering is implemented by adopting industry-specific terminology and expanding processes tailored to the unique requirements of the field. Furthermore, R. de Graaf et al. (2016) noted that SE is applied differently in the industry, with differences in the number of elements applied from SE. They indicate that there is not one standardized method for SE. Instead, it depends on aspects such as the company, its employees, and clients, which impact the level of SE application in projects.

For further analysis, this study primarily adopts the structure of the SE process as described by the US Department of Defense (DoD) (US Department of Defense, 2001), while also incorporating relevant elements from the technical processes described by INCOSE and essential handbooks and guidelines for SE in construction (Buck et al., 2023; R. de Graaf, 2020; INCOSE, 2015; Presland et al., 2018). This SE process has been studied and validated by different researchers, particularly in the context of the construction industry (R. S. de Graaf et al., 2017; R. de Graaf et al., 2016; Lynghaug et al., 2022). In addition, the SE process adopted in this research is focused on the technical aspects of SE as they align with the work of ProcessMinded. ProcessMinded has a ten-step approach for SE as follows: organize the project, analyze documents, analyze requirements, determine object tree, determine work packages and activities, allocate system requirements to objects, draw up a verification plan, perform verifications, carry out inspections, and validate and deliver. Thus, adopting the SE process as outlined by the DoD aligns with the study's goals and focus on ProcessMinded and keeps it in line with the broader application of SE in construction. The DoD established the MIL-STD-499B standard for SE with the following key steps: process input, requirements analysis, functional analysis and allocation, design synthesis, verification and validation. The iterative nature of the SE process is considered in the analysis later for integration with circular economy principles, including the requirements and design loops, which involve revisiting and refining requirements

and design based on feedback loops. Furthermore, the Dutch construction industry has developed its guidelines and framework for SE as a reference for the Dutch construction industry (Buck et al., 2023). This guideline by the leading Dutch organizations in the industry addresses the same SE process aligning with the DoD. In addition, according to Buck et al. (2023), SE applies to all contract types and supports the structured development of systems in the building and civil engineering sector. Thus, this research does not focus on specific contract types for further analysis and assumes that contractors and clients work together from the outset. The following sub-sections provide an overview of the Vee model and then delve into each step of the SE process.

3.2 Systems Engineering Process and the Vee Model

Before explaining each step of the SE process, it is important to address the iterative nature of Systems Engineering across different levels of design detail in construction projects. The Systems Engineering process is applied throughout the project lifecycle, from initiative to demolition, focusing on the design phase. This design phase typically involves different levels of detail. At each level, the SE process is applied iteratively, refining the design and moving closer to realization (R. de Graaf, 2020; INCOSE, 2015). As the project progresses through these levels, the same fundamental SE principles are applied, but with increasing specificity and detail; this is shown in the Vee model; see Figure 2 below. However, this study keeps the discussion of the SE process generic except for the process input, as it is assumed to be the earliest phase as the project start. This approach allows for flexibility, as the process can be applied at different levels of detail depending on the project's needs and stage.



Figure 2: The Vee Model and the SE process adapted from: R. de Graaf (2020)

3.2.1 Process input (Project Start)

Project start in SE is the process input step and establishes the system's foundation. This process is described as the initiative phase by (US Department of Defense, 2001) and aligns with the earliest stages of starting a project, as shown in the Vee model above. This involves analyzing stakeholders and defining needs, goals, and objectives through techniques such as interviews, focus groups, and surveys. With consideration to Measures of Effectiveness (MoE), the system's environment, legal boundaries, and constraints to guarantee sufficient coverage of system complexities from the start. It involves gathering and documenting initial sets of stakeholders' requirements, wishes, and project constraints, additionally, documenting the analysis conducted in stakeholder communication documents or other documents for effective communication and to be checked in later stages (R. de Graaf, 2020; Forsberg & Mooz, 1992; Locatelli et al., 2014; US Department of Defense, 2001).

3.2.2 Requirements Analysis

Requirements analysis starts by taking the output of the project start step and analyzing these requirements further. It connects the SE process with external input sources, using stakeholder communication documents and analysis. Requirements analysis is an inquiry and resolution process that focuses on understanding the reason behind developing the system, customer expectations, user needs, system functions, compliance, and constraints. This process results in performance, technical, interface, and functional and non-functional requirements. These requirements are documented at the end of the process for use in the subsequent steps of the SE process and for effective verification and validation, serving as a reference and will be revisited with the requirements loop (INCOSE, 2015; US Department of Defense, 2001).

3.2.3 Functional Analysis and Allocation

The functional analysis start by taking the output of the requirements analysis as an input. The process deals with translating requirements into functions the system should perform, and involves breaking down the system into smaller, manageable functions and allocating these functions to specific components or objects (US Department of Defense, 2001). The project team decomposes functions to transform requirements into functionalities (Buede, 2000). This process only describes what should be done and not how to do it, providing a solution-neutral functions formed as actions with verbs and nouns. It is an implementation-independent process, which ensures that subsequent trade studies for design are unbiased toward pre-determined options (INCOSE, 2015; Kossiakoff et al., 2011). The process is connected to both the requirements analysis and design synthesis through iterative loops. Functions are refined throughout the process with the requirements and are revisited later to iterate and refine the design. Several tools can be used to facilitate this process, such as Functional Flow Block Diagrams (FFBD) or Function Analysis Systems Technique (FAST) diagrams to show sequence and relationships, Integration Definition for Function Modeling (IDEF0) diagrams to define process and data flow, Timeline analysis to link functions to time and present time critical functions, and requirements allocation sheets to keep track of allocated functions and for later verification and validation. Key considerations for this process includes identifying all functional interfaces and functional performance.

3.2.4 Design Synthesis

This process starts by taking the output of the functional analysis and allocation, beginning with the design efforts of the system and starting to think of solutions. The project team considers solutions that satisfy the required functionalities, developing and comparing alternative designs in trade studies to determine which performs best. This process considers modular designs by grouping components with singular functions, minimizing interdependence and enhancing cohesion, to facilitate development, testing, and later adjustments (US Department of Defense, 2001). Initially, a conceptual design is created, followed by iterative refinement through preliminary and detailed designs. Tools such as the Requirements Analysis Sheet (RAS), Concept Description Sheet (CDS), and Schematic Block Diagram (SBD) ensure performance traceability, while software tools aid in presenting drawings and documenting design efforts. Design synthesis employs System Modeling Language (SysML), Computer Aided Design (CAD), and simulation tools to facilitate the design process (Kossiakoff et al., 2011). Iterative feedback from testing and stakeholder reviews is used for refining the design, making this process dynamic, bridging functional requirements with physical implementation (Buede, 2000). The design synthesis process involves selecting and integrating technologies, components, and interfaces to realize functional requirements. Guided by the project team's expertise, it focuses on creating an efficient and feasible system design with consideration of RAMS (Reliability, Availability, Maintainability, and Safety), cost, and performance (Blanchard & Fabrycky, 2005; INCOSE, 2015). The iterative nature ensures that the design remains aligned with the functional needs and can adapt to new information or changes in the system.

3.2.5 Verification and Validation

Verification and validation (V&V) are the processes that ensure a system meets its specified requirements and performs its intended functions (INCOSE, 2015; US Department of Defense, 2001). Verification is linked to the requirements analysis step and involves confirming that the system components are built and integrated correctly according to the predefined specifications, the main purpose of this process is to make sure that "the system is built right" (INCOSE, 2015). The verification takes place at different points of time during the development of the system and checks if the requirements are met by comparing the outputs of each phase of the system's development lifecycle against the input requirements of that phase. Tools and methods such as inspections, reviews, walkthroughs, and testing (including unit tests, integration tests, system tests, and acceptance tests) are employed to achieve verification. Validation, on the other hand, is linked to the project start and assesses whether the final system fulfills its intended purpose and meets stakeholders' expectations in the operational environment. This process ensures that "the right system is built" by evaluating the system's performance in real-world conditions and scenarios (INCOSE, 2015). Validation activities include operational testing, user trials, simulations, and demonstrations to gather evidence that the system meets its overall goals and user expectations. These processes are iterative and intertwined with the system development lifecycle, ensuring continuous alignment with requirements and stakeholder needs. The iterative nature of V&V involves repeated cycles of checking and feedback, which allow for early detection of issues and continuous improvement and refinement of the system design.

4 Circular Economy Principles in Construction

This section explores the principles of the circular economy within the construction industry, offering an overview of key concepts and frameworks. In addition, the section discusses the four principles adopted in this study, supplemented with examples of their application in the construction industry. It sets the stage for examining the integration of these principles into the SE process.

4.1 Frameworks and Principles of Circular Economy in Construction

This study aims to develop a framework that incorporates key circularity principles into the SE process addressed in the previous section 3. The application of circular economy principles in construction is a rapidly evolving field, with various frameworks and strategies proposed by researchers and practitioners. The Ellen MacArthur Foundation's ReSOLVE framework, for instance, outlines six actions for businesses to adopt circular practices: Regenerate, Share, Optimize, Loop, Virtualize, and Exchange (Ellen Macarthur Foundation, 2015). This framework provides a broad perspective on circularity that extends beyond material use to include business models and value creation. Moreover, the Dutch Platform CB'23 has also contributed greatly to this field, developing a framework that focuses on key circular design strategies. These strategies include prevention, design for quality and maintenance, design for adaptability, design for disassembly and reusability, design with reused parts of constructions, design with secondary raw materials, and design with renewable raw materials (Platform CB'23, 2023). Their framework provides practical guidance for implementing circular design strategies in construction project.

In addition to these practical frameworks, Pomponi and Moncaster (2017) developed a research framework identifying six key dimensions for studying and implementing circularity in construction: economic, environmental, technological, societal, governmental, and behavioral. Their work highlights the multifaceted nature of circularity in construction, emphasizing the need for a systemic approach that considers the entire lifecycle of construction projects. This framework supports a deeper understanding of the various factors influencing circularity and underscores the importance of an interdisciplinary approach. While these research and practical frameworks offer valuable insights, this study adopts a focused approach based on four key principles: closing, narrowing, slowing, and regenerating the resource loop. This choice is informed by the work of N. Bocken et al. (2021), N. M. Bocken et al. (2016), Cetin et al. (2021), and Konietzko et al. (2020) who have synthesized various circularity strategies into these four fundamental principles. The selection of these four principles is justified by several factors. **First**, these principles align closely with the construction lifecycle, addressing key stages from material sourcing to end-of-life considerations. They provide clear guidance on resource flows and lifecycle thinking, which are essential for effective integration with the SE process. Second, these principles have been successfully applied to study circularity within a digital built environment, demonstrating their relevance and applicability in addressing circularity challenges in the construction industry (Cetin et al., 2021). Third, these principles offer a simplified yet comprehensive approach, potentially encompassing strategies from both the Ellen MacArthur Foundation's ReSOLVE framework and Platform CB'23's design

strategies. Lastly, these four principles can account for the repercussions regarding the six key dimensions identified by Pomponi and Moncaster (economic, environmental, technological, societal, governmental, and behavioral) (Pomponi & Moncaster, 2017). This balanced approach could help develop a framework for integrating circularity principles into the SE process in construction.

4.2 The Four Circular Economy Principles

This study adopts four key circularity principles as outlined by N. Bocken et al. (2021), N. M. Bocken et al. (2016), and Konietzko et al. (2020) and further developed by Çetin et al. (2021): closing, narrowing, slowing, and regenerating the resource loop. These principles form the foundation for integrating circularity into the SE process for construction projects. Figure 3 illustrates these principles and their relationship to resource flows in a circular economy.



Figure 3: The Four Circularity Principles in Resource Flow Adapted From: Konietzko et al. (2020)

As shown in Figure 3, each principle plays a specific role in creating a circular flow of resources. The dotted line completing the circle represents closing, emphasizing material reuse and recycling. The thickest part of the circle illustrates narrowing, focusing on minimizing initial resource use. The wavy line at the top represents slowing, aiming to extend the lifespans of resources. Lastly, the plant icon indicates regenerating, representing the use of renewable resources and regenerative practices. The following subsections will explore these principles in depth and examine their application in construction projects.

4.2.1 Closing the Resource Loop (Use Again)

This principle emphasizes including secondary materials in new projects and ensuring that the resources used can be reused or recycled at the end of life. This principle focuses on minimizing waste by ensuring that end-of-life materials are reintegrated into the construction process rather than being disposed of as waste. An example of the application of this principle is the use of recycled materials in resin mortars, which demonstrates how old materials can be reused in new construction projects to reduce waste and contribute to closing the resource loop (Debska et al., 2024). In addition, this principle manifests in construction through key strategies such as designing for disassembly and material recovery, demonstrating how design choices significantly impact the potential for material recovery at the end of life of construction projects (Akinade, Oyedele, Ajayi, et al., 2017) Incorporating recycled or reclaimed materials into new projects is another strategy for closing the loop. However, it requires careful consideration of other factors such as sorting, processing, and proper reintegration of these recovered resources (Gálvez-Martos et al., 2018). On the same note, the "Building as Material Banks" (BAMB) project exemplifies the practical application of this principle by viewing buildings as temporary storage for materials, facilitating their future reuse (European Energy Innovation, 2017).

4.2.2 Narrowing the Resource Loop (Use Less)

Narrowing the resource loop focuses on resource efficiency, aiming to use fewer resources to achieve the same output. In construction, this principle is realized through various strategies that focus on reducing material use without compromising functionality or performance, such as longer-lasting products, modularisation and remanufacturing, and designing products with less material (Allwood et al., 2011). In addition, narrowing the resource loop includes utilizing advancements in technology and design to reduce the amount of raw materials and energy consumed in construction projects. *For example*, the use of 3D-printed clay-based mortars with innovative structures shows how efficient design can reduce resource consumption (Peng et al., 2024). Besides, better coordination among project stakeholders and accurate material estimates, which can be utilized with the use of BIM, could potentially lead to waste minimization (Liu et al., 2015).

4.2.3 Slowing the Resource Loop (Use Longer)

The principle of slowing the resource loop aims to extend the lifespan of materials and reduce the need for new resources. This principle is particularly relevant in construction, given the long-term nature of built assets. With the aim to use resources longer, designing for durability, adaptability, and flexibility are key strategies. *For example*, using durable and high-performance materials reduces the frequency of replacements during maintenance, resulting in lower resource consumption over the lifecycle of a project (Kibert, 2016). Additionally, planning for maintenance and repair are essential aspects for slowing the resource loop (Moraga et al., 2019).

4.2.4 Regenerating the Resource Loop (Make Clean)

This principle focuses on applying practices that positively impact the environment. This includes using materials and methods that help restore and regenerate the natural environment rather than depleting it. *For example*, organic wastes can be used for insulation in some construction projects (Indwar & Titiksh, 2024). The incorporation of biobased and renewable materials represents a key strategy in this context, offering opportunities for more circular material choices. The potential for materials that reduce environmental impact and contribute positively to ecosystems has been explored through biotechnologies and bioinspired materials in construction (Pacheco-Torgal & Labrincha, 2013). Moreover, implementing green infrastructure and nature-based solutions is another aspect of regenerating the resource loop (Maes & Jacobs, 2017)

5 Circular Systems Engineering Process

This section integrates the four circularity principles addressed earlier with the SE process. To integrate the four circularity principles into the SE process effectively, this study first analyzes the literature emerging from both paradigms and their intersections. Each step of the SE process is examined separately to identify potential areas for incorporating the four circularity principles. This critical assessment serves the development of a framework focusing on three primary elements: *inputs, actions, and outputs* within each step and for each circularity principle. The focus on the preceding three elements is grounded in the fact that each step is viewed as a process in itself. According to ISO/IEC/IEEE 15288 (2015), processes require clearly defined inputs, a set of actions or activities, and results in outputs. These elements are well-established in systems engineering processes (US Department of Defense, 2001), corresponding to the basic flow of processes: what goes in, what happens, and what comes out. The three elements are explained below:

- **Inputs** are the necessary resources and information that initiate each process step. By clearly identifying inputs, the framework ensures that all required elements are in place to proceed effectively.
- Actions are the specific activities undertaken within each process, considering circularity. These actions transform inputs into outputs.
- **Outputs** are the results that emerge from the actions taken during each process step. Clearly defined outputs allow for the process as a whole to proceed and enable evaluation of the integration's success and effectiveness.

While the framework focuses on inputs, actions, and outputs, the analysis in the following subsections expands to address challenges, opportunities, and potential solutions for integrating circularity principles with the SE process. This integration is presented in the framework in **Table 1** and supported by practical examples for certain actions shown in **Table 2** in section 5.6 is the result of a comprehensive synthesis of the literature. This comprehensive analysis forms the theoretical foundation and main reference for the framework. For clarity and practicality, the framework itself is structured around the core elements addressed above per SE step and distinguishes between the four circularity principles, providing a simplified, systematic approach for integrating circularity principles into the SE process.

Throughout the analysis, the term 'project team' is used to refer to the engineers and other professionals directly responsible for planning, designing, and executing the construction project, with systems engineers serving as the primary point of contact and main audience for this research. This inclusive terminology reflects the collaborative nature of integrating circularity into the SE process. The following subsections analyze each step of the SE process separately, providing the theoretical underpinning for the framework. The framework presented at the end will guide ProcessMinded in integrating circularity principles and form the core of this research.

5.1 Project Start

This section focuses on integrating the four circularity principles into the process input (project start) step of the SE process, providing an analysis of possible ways of integration and a theoretical foundation for developing a framework for integration. The project start is highlighted by the US Department of Defense (2001) as the process input, and it is viewed in this analysis as a distinct process due to its importance for integrating circularity principles as it paves the way for the succeeding steps. As explained before in section 3.2.1, the **inputs** of this step are initial demands and needs, systems environment and constraints, and measures of effectiveness (US Department of Defense, 2001). These inputs are considered to be the inputs from the initiative phase, coming from the earliest phase of a project. At this step, the inputs are the same for the four circularity principles: close, narrow, slow, and regenerate. However, if the SE process moves to the next level of detail, then the inputs will include previous development efforts.

At this step, setting clear circularity goals is essential, and equally important is the collaboration of stakeholders toward these goals (N. M. Bocken et al., 2016; Henderson et al., 2019; Kraaijenhagen et al., 2016). Setting circularity goals early in a project helps with decision-making, saving costs, and providing economic benefits throughout the project's lifecycle (Bragança et al., 2014). In addition, early-phase planning is essential for integrating circular economy principles (Sanchez & Haas, 2018). It helps guide decisions such as choosing between green-field construction and adaptive reuse, planning for closed-loop cycle construction, and optimizing the benefits of adaptive reuse. While SE is flexible and can adapt to changes (Dove & LaBarge, 2014), starting with clear circularity goals and involving experts from the outset is crucial to avoid costly adjustments later in construction projects (N. M. Bocken et al., 2014; N. M. Bocken et al., 2016; HM Treasury, 2013). Considering the four circularity principles, this translates to **actions** that concern setting targets for resource recovery, efficiency, longevity, and bio-based materials use; see **Table 1** for the related **actions**.

Furthermore, the project start in SE involves stakeholder analysis and defining objectives. However, conventional stakeholders like clients and designers often lack awareness of circularity principles (Adams et al., 2017). Therefore, it is necessary to *include circular*ity experts and educate stakeholders on circularity to foster collaboration and effectively leverage circular economy principles (Gerding et al., 2021). The inclusion of circularity experts represents a transition from linear to circular engagement and requires proactive collaboration, in-depth dialogues, training, and addressing knowledge gaps through consultants and internal expertise (Fobbe & Hilletofth, 2023). However, this inclusion of circularity experts can help in setting the right targets at the project start (Adams et al., 2017). For such collaboration, the use of interactive and dynamic tools, such as circular design dashboards, can provide stakeholders with insights regarding circularity within the project, which can help them make informed decisions and identify effective circular building measures (van der Zwaag et al., 2023). Using such dashboards and sharing information could increase transparency and stakeholder commitment to circular goals from the start. Similarly, the power and interests of stakeholders differ and can potentially influence the innovative efforts of other stakeholders to consider circularity (Tookey et al., 2011). Clients have significant power and could have less interest in some cases due to a lack of awareness or financial concerns (Adams et al., 2017; Ahmed et al.,

2023). The project team can work toward initiating a dialog that balances power, focusing on stakeholders who might have less power yet be of significant value for achieving a circular project (Bal et al., 2013). As such, the project team should prioritize circularity alongside conventional interests, considering non-human stakeholders such as nature, to achieve outcomes beyond usual standards (Phillips & Reichart, 2000; Senaratne et al., 2021; World Economic Forum, 2020).

This way of collaboration presents challenges, such as limited incentives, collaboration barriers, decision-making complexities, and implementation risks, as highlighted by Shooshtarian et al. (2024). They show a case study of a construction project using recycled content, revealing stakeholders differing views on circularity and stressing the need for effective communication and early collaboration to align on circularity goals. Given these complexities, SE can *facilitate a strategic narrative* for a shared vision and values to create a sense of ownership among stakeholders toward circularity goals (Henderson et al., 2019). Aligning to a strategic narrative and early engagement at the project start can help address and manage misalignment issues before project complexity increases. On a similar note, the Concept of Operations (ConOps) in SE is a high-level tool that can be deployed to help stakeholders align to a unified vision, facilitating a better communication of goals (Madni, 2015; Mostashari et al., 2012). ConOps can be used to describe a project's circularity vision and objectives at the start to help everyone understand what would be considered a success later on (ISO/IEC/IEEE 15288, 2015). For circularity, concrete indicators to measure progress are needed (PwC, 2019). A highlevel ConOps can define and communicate these goals to stakeholders, making it possible to measure progress. It can also help to involve stakeholders in the value creation and problem-solving process as a pragmatic and strategic engagement (Kujala et al., 2022). Additionally, it can be seen as a goal-oriented engagement with a focus on value and knowledge creation toward circular goals in a project (Oberholzer & Sachs, 2023). A stakeholder mindset shift towards viewing circularity as an intrinsic value can influence every decision and action, which is described as a leverage point for achieving circular SE by David et al. (2024). The project team can consider conducting an inventory analysis for material harvest and to identify possible local sources, such as existing projects that are put for deconstruction in the area (Blok, 2024; Bragança, 2019), which could help in setting targets for closing the resource loop. Similarly, identifying renewable resources within the project's local environment and conducting Preliminary Ecological Appraisal (PEA) (CIEEM, 2017) could improve the process of setting targets for regenerating the resource loop and could be used later in the process. The **actions** discussed above and their **outputs** within the four circularity principles are shown in **Table 1** below.

5.2 Requirements Analysis

This step starts with the input from the previous step, with goals and targets related to the four circularity principles, as well as an inventory analysis of local sources for reclaimed materials and renewable and bio-based sources. These inputs form the first set of requirements that the project team would work on analyzing. Ecodesign guidelines distinguish between generic and specific ecodesign requirements; generic requirements are related to overall environmental performance, and specific requirements are related to environmental aspects within a specific product (European Commission, 2009). Circularity requirements can be analyzed in a similar way to those explained by the ecodesign guidelines. Where generic requirements focus on lifecycle considerations within all system requirements, and specific requirements are explicit (e.g., full deconstruction and reclamation of building elements after a specified period). Generic requirements could result from integrating circularity principles and including targets for each circularity principle within the project, bringing new considerations for requirements analysis. Analyzing these requirements can be done to ensure they fulfill the circularity targets within the overall system's performance (Blyth & Worthington, 2002). In addition, they need to be analyzed with consideration of the project's environment to avoid compromising other environmental aspects. At the early stage of the project, it can be challenging to capture all generic and specific circularity requirements due to limited information and differing stakeholder's needs (Lehtinen et al., 2023). However, requirements analysis is a dynamic process where requirements are initially set based on available information and further refined as the project progresses (Jensen, 2011; Prins et al., 2014). Therefore, distinguishing between generic and specific requirements is considered as an **action** within the four circularity principles as it sets the ground for further analyzing every set of requirements with the right tools and involving the right people.

At this step, requirements are analyzed for circularity at the sub-system level; later, when the complete set of requirements is available, they can be combined to reflect the entire system's circularity performance. Frameworks like the Madaster Circularity Indicator, the CirculAbility Model, and the Measuring Circularity Guide emphasize measuring the reusability and recyclability of materials, evaluating material and energy flow, and the lifespan of materials (Heisel & Rau-Oberhuber, 2020; Inchainge, n.d.; Platform CB'23, 2022). Based on these frameworks and guidelines, requirements could be assessed for a closed-loop system by considering material recovery, EOL phase, and future reuse and recycling potential. EOL consideration could include aspects such as ease of disassembly, maintaining plans and documents, and reuse potential (Anastasiades et al., 2023). To narrow the loop within the project, the project team should consider analyzing requirements based on materials and energy consumption using LCA and ECI. For slowing the loop, analyzing requirements for longevity and repairability is essential. In addition, assessing possible bio-based and renewable alternatives for requirements is to move toward a regenerative project. See Table 1 for the actions within each principle. This integration includes other considerations; for example, the project team can involve demolition contractors to help analyse EOL requirements. According to Osaily et al. (2019), early demolition contractor involvement can be invaluable due to their experience regarding factors influencing the EOL of construction projects, helping formulate requirements that adhere to closing the loop. Similarly, technical experts can be included when needed for requirements related to the four circularity principles. However, adversarial relationships, increased costs, and time constraints might arise from such involvement (Osaily et al., 2022), which should be considered carefully by the project team. In addition, circularity requirements mandate openness to new and innovative ways of working, which could potentially cause adversarial relationships in the project (Rose & Manley, 2014). For that, the project team should consider managing relationships carefully in this step to guarantee a successful analysis of the requirements.

Furthermore, it is essential for the SE process that requirements are clearly documented to ensure traceability and clarity and to facilitate later verification and validation while remaining solution-neutral to minimize uncertainties and ensure they are actionable (Arayici et al., 2005; Kamara & Anumba, 2002; Wheeler, 2004). There are different frameworks to facilitate sufficient documentation of requirements (e.g., making them $SMART^{1}$) so they can be communicated and understood by everyone at different phases of the project (MannionMike & KeepenceBarry, 1995). However, measuring circularity is challenging due to the lack of standardized practices and the complexity of evaluating lifecycle impacts considering the uniqueness of construction projects (Incelli et al., 2023; Rahla et al., 2019). Several circularity indicators exist, but none can measure all aspects of circularity alone (Moraga et al., 2019). Therefore, using multiple indicators and metrics, such as the Environmental Cost Index (ECI/MKI), CO2 emissions reductions from Life Cycle Assessment (LCA), Material Circularity Index (MCI), and Material Flow Analysis (MFA), can help make circularity requirements measurable (González et al., 2021; UKGBC, 2019). The project team can make use of these existing tools to ensure the documentation of measurable requirements for each circularity principle. At the end of this step, all circularity requirements should be documented. These requirements can be integrated into a design brief and requirements verification traceability matrix (RVTM), which are common documents that are used within SE (INCOSE, 2015), with a distinction between generic and specific circularity requirements to proceed further in the SE process.

5.3 Functional Analysis and Allocation

Functional analysis and allocation start by taking the output of the requirements analysis step and analyzing them further for functionalities and allocating functions to objects. For circularity principles, the focus is on functionalities such as reusability, longevity, adaptability, and deconstructability, ensuring materials and components remain circular throughout their lifecycle (Adams et al., 2017; Hubmann & van Maaren, 2022). Circular thinking emphasizes considering the entire system and examining components and their interfaces separately, similar to functional thinking in SE. Aligning with this is the concept of approaching buildings as layers (site, structure, skin, services, space plan, and stuff) (Brand, 1995). The layers concept can boost circularity by allowing the project team to address each element with distinct rules to integrate circularity (Cheshire, 2016; Dams et al., 2021). In SE, it is up to the project team to decide on the decomposition of functions (Buede, 2000). Using the layers concept can help estimate the lifecycle of different components to group them together and move from higher-level to lower-level functions. Similar to the previous steps of the SE process, the project team might involve circularity experts for assigning functions to objects as this needs further multidisciplinary collaboration (Price, 1985).

As discussed in section 5.2, resulting from integrating circularity, there will be generic and specific circularity requirements. On the one hand, specific circularity requirements could result in basic functions such as, but not limited to, reusing components and recycling end-of-life products. These functions can be further decomposed into lower-level functions and then allocated to objects. The use of the FAST diagram can help break down functions related to specific circularity requirements following the how-why logic and moving to lower-level functions. On the other hand, generic circularity requirements would result in qualities that describe the desired characteristics of the system, such as reusability, efficiency, durability, and eco-friendliness, which can be treated as support-

¹SMART stands for Specific, Measurable, Achievable, Realistic, and Timed.

ing functions similar to the RAMS (Reliability, Availability, Maintainability, and Safety) (Buck et al., 2023; R. de Graaf, 2020). Techniques such as Fault Tree Analysis (FTA), Event Tree Analysis (ETA), and Failure Mode and Effect Criticality Analysis (FMECA) are used for RAMS analysis (Al-Jibouri & Ogink, 2009). These techniques can similarly be used to identify and evaluate failure modes and performance related to reusability, efficiency, durability, and environmental impacts, ensuring these circularity requirements are covered within functions, see **Table 1** for related **actions**. For example, assessing the reusability of functions can provide insights regarding their value at the end of life (Platform CB'23, 2022, 2023). Based on these assessments, functions can be grouped together and allocated to modular components that are easy to disassemble and reuse, considering the principle of closing the resource loop. The same can be applied to narrowing, slowing, and regenerating the loop. The project team can decide their design ratios when grouping and allocating these functions to objects (Buede, 2000).

Furthermore, circularity principles and the functional analysis and allocation step have overlapping concepts, with this integration, their strengths could complement each other and enhance the overall circularity of the process. This step of the SE process takes a holistic approach through managing functions and their logical flow, emphasizing performance and functionality, yet, neglecting material flow. Conversely, circular economy principles constantly focus on material inputs and outputs but could lack a systemic view of functionality for operation and use (Hubmann & van Maaren, 2022). Additionally, functional analysis and allocation emphasize output thinking (what the system should achieve), while circularity principles focus on input and output thinking in terms of resources, like minimizing the system's waste output by managing inputs (Rahla et al., 2021). Thus, incorporating material flow thinking such as reusability, efficacy, and durability in this process can better align it with circular economy principles. In addition, physical and functional interface management can ensure the system's elements and materials interact efficiently for resource optimization and ease of assembly and disassembly. Managing interfaces, in this case, means ensuring that components and materials interact seamlessly, considering strategies such as reversibility, simplicity, and speed (Vandenbroucke, 2016). An Interface Control Document (ICD) can be used to specify physical and functional interfaces in the system, actions related to interfaces are addressed in Table 1 below. In addition to the interface control document, the **outputs** of this step include the object tree with the different grouped objects based on the circularity principle considered and the requirements allocation sheet (RAS) for traceability and later use in the design synthesis step. Lastly, it is worth noting that incorporating circularity principles may impose immediate constraints on the design space (Qimen, 2021). Yet, this step of SE can still be carried out as a solution-neutral process, allowing for broader exploration of the design space in the next SE process.

5.4 Design Synthesis

The design synthesis step starts with the outputs from the previous step representing the functional architecture of the system and proceeds to develop design alternatives based on that. For circular design, the focus is on considering the whole life cycle and designing projects where materials are optimized and can be reused, recycled, or repurposed at the end of their lifecycle (Dewagoda et al., 2022; Schützenhofer et al., 2022). Designing for circularity in the construction industry requires a systemic and integral approach, consid-

ering both the lifecycle of each component individually and their interactions within the whole system (van Stijn & Gruis, 2020). In addition, the selection of materials, the modularity of design, and ease of assembly and disassembly for future reuse are prioritized during design (Dams et al., 2021; Minunno et al., 2020; Rahla et al., 2021). To design for circularity, there are various strategies, such as the use of standardized components that can be easily assembled and disassembled, promoting adaptability and prolonging the lifecycle of materials (Eberhardt et al., 2022).

In construction projects, many components are designed to be interlocked together, making it challenging to separate them when needed. These components are usually connected and depend on each other for stability and functionality. However, this step of the SE process focuses on developing systems that consist of subsystems with careful management of functional and physical interfaces, which could help overcome this issue. From the efforts of the previous step, as explained in section 5.3, the design process can proceed further with a design solution focusing on functional autonomy and facilitating modularity. The design synthesis in SE aligns with the circularity principle for managing interdependencies and interfaces. This alignment can advance circularity in construction projects by easing assembly, disassembly, repurposing, and maintenance activities. Vandervaeren et al. (2022) show through material flow analysis that when considering the interdependencies between building parts and assessing the environmental impacts of demountable buildings, a reduction in material consumption and waste occurs. Two examples demonstrate this as shown by Guy and Ciarimboli (2005); first, the Open House project by Bensonwood Homes, where they used a 3D-design approach with prefabricated elements, treating the building components as independent subsystems, which resulted in enhanced maintainability and future adaptability. Second, the R 128 house showcases how modular prefabricated systems can be designed for dismantling, ensuring that materials can be reclaimed for reuse or recycling (Guy & Ciarimboli, 2005). To take actions within the design synthesis step, the project team can consider designing for disassembly and reassembly for a closed loop system (Platform CB'23, 2023). With a focus on whole lifecycle considerations and demountable interfacing of components. For narrowing the loop, it is evident that a focus on minimal material use is needed. This can be achieved by considering optimizing geometries and lightweight structures (Durmisevic et al., 2019; Tingley, 2014). While slowing the resource loop, the project team can consider durable options within the design, considering maintenance and upgrade strategies for future changes (Eberhardt et al., 2022). For a regenerative design, the project team should incorporate renewable and bio-based alternatives when possible (Attia, 2018), utilizing inventory analysis and preliminary ecological appraisal conducted earlier in the SE process as highlighted in section 5.1. These actions are listed in **Table 1** below. However, it is important to note that these actions within the four principles complement each other, and the project team can include what is possible based on the project nature and evaluate design alternatives in trade studies for the optimal option.

Furthermore, the project team should provide disassembly plans with a sequence of activities for disassembly alongside construction, operations, and maintenance plans (Tingley, 2014). These plans will be needed at the end-of-life and can be used for repurposing or replacing certain parts during the lifecycle of the project (Bouyarmane et al., 2020). Early in the design process, end-of-life considerations such as disassemblability assessment can help generate and compare alternatives for design trade studies (Desai & Mital, 2003; Zhu et al., 2020). In addition, materials passports can be used to facilitate documenting information and tracking materials (Göswein et al., 2022; Heinrich & Lang, 2019). The use of material passports can be utilized for closing the loop through the traceability of materials for reclamation at the end-of-life and also for slowing the resource loop by facilitating maintenance tasks through extracting important information needed easily. In addition, BIM can be utilized to simulate certain circularity concerns, such as assembly and disassembly workflows, easy accesses for maintenance, and detect clashes for feasible and executable designs (Akinade, Oyedele, Omoteso, et al., 2017; Azhar et al., 2012; HM Government, 2013; Van Den Berg et al., 2021). The **outputs** of this step are the design alternatives based on the level of detail of the SE process, with the consideration of the four circularity principles. The outputs could be integrated to complement and reinforce each other or be compared in trade-off studies to find the optimal circular design solution. See more for the related **actions** and **outputs** in **Table 1** below.

5.5 Verification and Validation

The verification and validation process could take place at different times of the SE process for iteration, as discussed in section 3.2.5. In this section, the focus is on verifying and validating the circularity aspects of the design once the design synthesis step is completed. This narrowed focus on verification and validation makes it easier to address issues related to integrating circularity, ensuring that key elements are thoroughly evaluated for the integration.

In the CE, Conformity Assessment (CA) ensures that products and processes meet circularity requirements with similar concepts to the verification and validation process in SE (ISO, 2022; Vehring, 2023). Both aim to systematically assess system compliance with specific requirements, utilizing tools to ensure transparency and traceability. Verification has been deployed for checking compliance with environmental impact reports in the CHSTP project (O. M. Hoehne, 2012). In the same way, when circularity requirements are considered in the system, verification can incorporate checks for circular design compliance, such as ensuring that components and materials are verified for their reusability, recyclability, and environmental impact, see the related **actions** for each principle in **Table 1**. For instance, materials can be inspected and tested for their lifecycle performance and potential for future reuse based on the system's requirements.

Validation, on the other hand, could be expanded to include circularity metrics, assessing whether the system's operational performance aligns with circular goals and targets set earlier for the project as discussed in section 5.1. This could include validating the system's effectiveness in minimizing waste and checking the disassemblability and recyclability of components for EOL. Moreover, Life Cycle Assessment (LCA), Material Flow Analysis (MFA), and Environmental Impact Assessments (EIA) can be used for validation purposes (Antwi-Afari et al., 2023). Nevertheless, the validation process will rely mainly on simulations and estimations as many of the circularity considerations are reflected in the future. Tools discussed for assessing functional performance in section 5.3 similar to RAMS could also be employed for validation based on future scenario analysis. For this integration, validation is focused on the goals and targets set at the project start, **actions** are derived accordingly as shown in **Table 1**.

However, traditional methods of verification and validation might require some changes when considering the integration with circularity principles, as these principles might require to be checked based on future scenarios. For example, checking if parts of the system meet disassembly requirements might require considering the disassembly scenario of the system as a whole. The use of BIM could help the project team conduct checks for verification and validation considering circularity (Andrich et al., 2022; van der Zwaag et al., 2023). In addition, sustainable design standards such as BREEAM, LEED, and DGNB can be used for the verification and validation of circular aspects (Pelsmakers et al., 2022; RIBAJ, 2022). This necessitates using models and simulations that fit circularity requirements, as stated by INCOSE (2021) in their 2035 vision for SE that verification and validation will require new methods and models in the future. Based on the way the project team finds it suitable to conduct verification and validation, the **outputs** of this step should include the verification and validation reports, verification matrix, and conformity assessment.

5.6 Framework and Examples

The following tables present the framework for integrating circular economy principles into the SE process. Table 1 outlines the key inputs, actions, and outputs for each SE step across the four circularity principles, while Table 2 provides practical examples for the actions with indices. The numbers in parentheses next to some actions in Table 1 correspond to the indices of practical examples provided in Table 2, offering concrete illustrations of how these actions might be implemented in real-world scenarios.

5.6.1 Circularity Integrated SE Process Framework

SE/CE	Close	Narrow	Slow	Regenerate		
	Project Start					
Inputs	Initial demands and needs System environment and constraints Measures of Effectiveness (MOE)	Initial demands and needs System environment and constraints Measures of Effectiveness (MOE)	Initial demands and needs System environment and constraints Measures of Effectiveness (MOE)	Initial demands and needs System environment and constraints Measures of Effectiveness (MOE)		
Actions	Include circularity experts Set targets for resource recovery for the EOL (e.g., 70% recyclability of materials) (1). Identify possible local sources for material harvest (2).	Include circularity experts Set targets for resource efficiency targets (e.g., 25% reduction in materials use and energy consumption) (3).	Include circularity experts Set targets for longevity, maintainability, and repairability.	Include circularity experts Set targets for bio-based materials use. Identify renewable resource opportunities within the local environment (4). Conduct Preliminary Ecological Appraisal (PEA) (5).		
Outputs	Goals and targets for early materials re- covery and the EOL. Materials inventory (e.g., secondary and reclaimed materials availability from local sources)	Goals and targets for material reduction	Goals and targets for longevity, maintain- ability, and repairability	Goals and targets for bio-based materials use. Inventory of local bio-based sources. Preliminary Ecological Appraisal (PEA)		
	Requirements Analysis					
Inputs	Output of the previous step	Output of the previous step	Output of the previous step	Output of the previous step		
Actions	Distinguish between specific and generic circularity requirements. Analyze requirements considering EOL phase (e.g., potential reuse and recycling using MFA and LCA) (6). Analyze requirements for opportunities to include secondary or reclaimed materials (7).	Distinguish between specific and generic circularity requirements. Analyze requirements based on materials and energy consumption (e.g., using LCA and ECI)(6).	Distinguish between explicit and general circularity requirements. Analyze requirements for longevity, main- tainability, and repairability (e.g., using LCEA) (6). Determine the expected lifespan of require- ments (6).	Distinguish between specific and generic circularity requirements. Assess possible bio-based, eco-friendly al- ternatives for a requirement.		

 Table 1: Framework for Integrating Circularity Principles into the Systems Engineering

 Process

SE/CE	Close	Narrow	Slow	Regenerate
Outputs	EOL requirements (e.g., integrated into the design brief and RVTM). List of requirements for secondary and re- claimed materials use.	Resource efficiency requirements (e.g., in- tegrated into a design brief and RVTM).	List of longevity and adaptability require- ments (e.g., integrated into the design brief and RVTM). Maintenance and repair strategy docu- ment.	Bio-based alternatives integrated into re- quirements (e.g., integrated into the design brief and RVTM).
		Functional Analysis and	d Allocation	
Inputs	Output of the previous step	Output of the previous step	Output of the previous step	Output of the previous step
Actions	Include specific circularity requirements as basic functions and generic as supportive functions (8). Consider reusability as a supporting func- tion when assessing functional perfor- mance. Allocate functions to modular components for easy disassembly (10). Assess and manage physical interfaces for seamless interaction (9).	Include specific circularity requirements as basic functions and generic as supportive functions (8). Consider efficiency as a supporting func- tion when assessing functional perfor- mance. Identify opportunities for multifunctional components. Allocate functions to optimize material use (e.g., multifunctional components) (10). Assess and manage functional interfaces for resource optimization.	Include specific circularity requirements as basic functions and generic as supportive functions (8). Consider durability as a supporting func- tion when assessing functional perfor- mance. Assess and manage physical and functional interfaces for easy repair and replacement.	Include specific circularity requirements as basic functions and generic as supportive functions (8). Consider eco-friendliness as a supporting function when assessing functional perfor- mance. Assess functions for their environmental impact. Allocate functions considering possible symbiotic relationships with the local en- vironment.
Outputs	Object tree, including modular compo- nents. Requirements Allocation Sheet (RAS). Interface Control Document (ICD).	Object tree, including multifunctional components. Requirements Allocation Sheet (RAS). Interface Control Document (ICD).	Object tree, including upgradeable and long-lasting components. Requirements Allocation Sheet (RAS).	Object tree, including the environmental performance of components. Requirements Allocation Sheet (RAS).
		Design Synthe	esis	
Input	Output of the previous step	Output of the previous step	Output of the previous step	Output of the previous step
Action	Design for disassembly and reassembly (e.g., reusable and recyclable materials, re- versible connections, and modular compo- nents) (10). Utilize material passports for traceability of materials (e.g., for EOL recovery and reuse) (11). Document disassembly plans (e.g., using BIM to model the sequence of deconstruc- tion activities and store information for ca- pabilities needed at the EOL).	Design for minimal material use (e.g., lightweight structures, optimized geome- tries, and prefabrication techniques) (12). Use parametric design tools and standard- ize components. Utilize BIM models for precision in mate- rial quantity surveying.	Design for durability and easy mainte- nance (e.g., accessible systems, robust ma- terials) (13). Incorporate upgrade pathways into the de- sign (13). Utilize material passports and BIM models for maintenance tasks.	Design using renewable and bio-based materials (e.g., green infrastructure and nature-based solutions) (12). Apply bioregional design methods
Output	Design alternatives based on the level of de- tail (including EOL and disassembly plans, the use of reclaimed materials)	Resource-efficient design alternatives based on the level of detail	Long-lasting and adaptable design alterna- tives (including maintenance and upgrade plans)	Regenerative design alternatives (includ- ing bio-based materials and green infras- tructure)
		Verification & Val	idation	
Inputs	Output of the previous step	Output of the previous step	Output of the previous step	Output of the previous step

SE/CE	Close	Narrow	Slow	Regenerate	
Actions	Verify EOL requirements (e.g., recyclabil- ity and recoverability of design choices) (14). Validate achieving material recovery tar- gets (e.g., using simulation software for future scenarios and BIM for disassembly workflow).	Verify design meets resource efficiency re- quirements (e.g., test resource efficiency performance) (15). Validate achieving material reduction tar- gets.	Verify that designs meet longevity and adaptability targets (e.g., conduct durabil- ity and adaptability testing) (16). Validate longevity, maintainability, and re- pairability targets (e.g., aging tests, and using BIM and simulation to check the feasibility of repair and maintenance pro- cesses).	Verify design meet regenerative require- ments (17). Validate the inclusion of bio-based mate- rial and achieving environmental impact targets (e.g., using EIA and LCA tools).	
OutputsVerification and Validation reports (e.g., EOL recyclability assessment, disassembly simulation results).Verification and Validation re resource efficiency test result consumption reduction analysi Verification matrix (e.g., showing compli- ance for material recovery requirements). Conformity Assessment (CA) (e.g., report on adherence to recycling and reuse tar- gets).Verification and Validation re resource efficiency test result consumption reduction analysi Verification matrix (e.g., show ance for resource efficiency req Conformity Assessment (CA) (e.g., report on adherence to recycling and reuse tar- gets).		Verification and Validation reports (e.g., resource efficiency test results, material consumption reduction analysis). Verification matrix (e.g., showing compli- ance for resource efficiency requirements). Conformity Assessment (CA) (e.g., report on adherence to material and energy effi- ciency targets).	Verification and Validation reports (e.g., durability test outcomes, maintenance and repair feasibility assessments). Verification matrix (e.g., showing compli- ance for longevity and adaptability re- quirements). Conformity Assessment (CA) (e.g., report on adherence to product lifetime extension targets).	Verification and Validation reports (e.g., bio-based material compliance checks, en- vironmental impact assessments). Verification matrix (e.g., showing compli- ance for regenerative design requirements). Conformity Assessment (CA) (e.g., re- port on adherence to regenerative and eco- friendly material targets).	
Note: Numbers between parentheses next to some actions are indices for examples from practice for these actions shown in Table 2.					

5.6.2 Examples from Practice

Table 2 below presents practical examples from practice that illustrate the implementation of seventeen actions from the framework, these examples provide practical insights and further support for the suggested actions in the framework.

Action Index	Example	Source
1	The Philips Circular Lighting project set a target to achieve 80% recyclability for their LED lighting products.	(Philips, 2015)
2	The Dutch company New Horizon Urban Mining identifies and harvests reusable mate- rials from buildings scheduled for demolition within a certain radius of their project sites.	(New Horizon, n.d.)
3	The Empire State Building retrofit project set a target to reduce energy consumption by 38%.	(Buildings, 2012)
4	The Bullitt Center identified opportunities to use local rainwater and geothermal for en- ergy, setting a target to be net-zero water and energy.	(B. Pena, 2014)
5	The HS2 railway project in the UK con- ducted PEAs along the entire route to iden- tify key ecological features.	(GOV.UK, 2022)
6	The new update of EU's Ecodesign for Sus- tainable Products Regulation requires anal- ysis of product requirements with considera- tion for durability, reparability, recyclability, ease of end-of-life disassembly and reuse, and recycled content.	(Ecochain, 2024; EUR-Lex, 2022)
7	The Dutch company Stonecycling analyzes building material requirements to incorpo- rate up to 60% waste-based content in their bricks.	(Treggiden, 2020)
8	In the design of Philips' pay-per-lux light- ing system, the basic function is providing light (with specific circularity requirements like energy efficiency), while supportive func- tions include recyclability and upgradability.	(Ellen Macarthur Founda- tion, 2022)
9	In the guideline of the EU-funded Buildings as Material Banks (BAMB) project, build- ing components are assessed for interfaces to enable future disassembly and reuse.	(Cornet et al., 2016)

Table 2: Supporting examples for the actions in the framework

10	The Dutch company Superuse Studios de- signs buildings using reclaimed materials, al- locating functions to these materials based on their inherent properties demonstrating how to close and narrow the resource loop through reuse and optimization.	(Baldwin, 2021)
11	EPEA and Madaster creates digital material passports for buildings, documenting all ma- terials used in construction to facilitate fu- ture reuse and recycling.	(Madaster, 2022)
12	The MycoTree project by Karlsruhe Insti- tute of Technology and ETH Zurich cre- ated a load-bearing structure using mycelium (mushroom roots) combined with bamboo, showing the potential of both bio-based ma- terials and lightweight structures.	(World Architects, 2017)
13	The Edge office building in Amsterdam is de- signed with accessible systems and durable materials, facilitating easy maintenance and upgrades.	(ArchDaily, 2016)
14	The DGNB (German Sustainable Building Council) certification system includes crite- ria for verifying the recyclability and recov- erability of building materials at EOL.	(DGNB, n.d.)
15	The Embodied Carbon in Construction Cal- culator (EC3) tool helps verify the resource efficiency of designs by comparing the em- bodied carbon of different material choices.	(Carbon Leadership Forum, 2023)
16	The BRE (Building Research Establishment) in the UK conducts durability testing on construction materials and systems to verify their longevity and long term impact.	(BRE, n.d.)
17	The Living Building Challenge have certifi- cation program to verify designs meet regen- erative requirements, such as net-positive en- ergy.	(Living Future, n.d.)

6 Framework Validation

This section outlines the framework validation process, starting with an overview of the procedure followed and details about interviewees. It then presents insights for each SE step of the framework. The validation focuses primarily on qualitative insights, supplemented by quantitative data in the form of applicability scores for integrating circular economy principles at each SE step.

6.1 Procedure and Interviewee Details

The framework developed in this study aims to incorporate four circular economy principles into the SE process. This was validated through semi-structured interviews with systems engineers from ProcessMinded as explained in section 2.3. The validation seeks to assess the framework and identify its strengths and potential improvements. In addition, the framework validation focuses on the actions that a system engineer should consider within every step of the SE process and addresses the applicability of these actions on a score from 1 to 5, with 5 being highly applicable and 1 not applicable at all. Information about the interviewees' roles and years of experience can be seen in Table 3 below.

Participant code	Expertise	Years of Experience
P1	Sustainability advisor, previously a systems engineer	13
P2	Systems Engineer	15
P3	Sustainability advisor, previously a systems engineer	17
P4	Process manager previously a Systems Engineer	8
P5	Project manager, previously a Systems Engineer	12

Table 3: Interviewees code, role, and experience.

6.2 Validation Insights Per SE Step

As the framework was presented to interviewees per SE step, the validation insights are presented in a similar way. These insights are then summarized in Table 4 at the end of this section, which shows the strengths and potential points of improvements for the actions per SE step in the framework and the average score of applicability of each step rated by the five participants.

6.2.1 Circularity Integrated Project Start

The Project Start step, with an average applicability rating of 4.2, was generally well-received. However, the interviews revealed an interesting tension between the perceived importance of this step and the ambiguity surrounding the systems engineer's role. **P3** raised an important question about responsibility:

"It is really hard to say if these actions are a responsibility of the system engineer. Or does it come from the ambitions of the project."

This points to a potential need for assigning roles and responsibilities when implementing circularity principles, especially at the project's inception.

An interesting suggestion came from P5 regarding ownership of circularity goals:

"Try to acknowledge the targets to a person to making them responsible to create ownership that you can need for the next actions."

This idea supports the framework's emphasis on setting clear circularity targets at the start yet highlights that personal accountability for circularity targets could be a powerful tool for ensuring these principles are carried through the entire project effectively.

6.2.2 Circularity Integrated Requirements Analysis

The Requirements Analysis step received an average applicability rating of 4, reflecting a mix of optimism and concern among the interviewees. A key insight from this phase was the challenge of translating broad circularity goals into specific, actionable requirements. **P2** observation that

"Nothing was explicit enough, I guess" ... "especially in civil engineering, It is such a new topic, they are not used to thinking at least in the setting of requirements for this, they are not used to it"

This highlights a common struggle in implementing circularity principles. This suggests that the framework is beneficial in providing more guidance on how to make circularity requirements more concrete and measurable. Furthermore, an interesting perspective came from $\mathbf{P1}$, who pointed out a potential conflict between different circularity principles:

"If I have an element with a short lifetime and I have to replace it several times, but it is circular because it's made of biobased, I do not know that it might be a better solution than an oil-based product, but last maybe longer."

This insight underscores the complexity of circularity and the need for balanced decisionmaking in requirements analysis within the different circular targets. These observations validate the framework's approach to distinguishing between specific and general circularity requirements. The moderate applicability rating (4) compared to the other steps reflects the inherent challenges in this step but also suggests that the framework provides a viable starting point for translating broad circularity requirements into concrete requirements.

6.2.3 Circularity Integrated Functional Analysis and Allocation

With an average applicability rating of 4, this step was viewed positively by most interviewees. The discussions around this step revealed a strong focus on the practical aspects of integrating circularity principles within the systems engineering process. **P2** provided valuable insight into the role of the systems engineer in this phase:

"These are real actions for the systems engineer only at first, of course, then, with the support of all the technical engineers again."

This comment validates the framework's approach of integrating circularity considerations into the core systems engineering processes while also recognizing the need for collaboration with those who have the needed technical expertise. An important aspect of this step is the management of interfaces, which is crucial for integrating circularity principles. **P5** highlighted this:

"You can say prioritize and manage the conflicts, manage their interfaces."

This observation aligns with the framework's emphasis on interface management as a key aspect of functional analysis and allocation for integrating circularity. **P3** underscored the importance of this phase, stating:

"I think in this phase it is important to make the interfaces between the materials clear for the reversibility, so if they can release the walls from the floor for example."

The insights from this step strongly validate the framework's approach to integrating circularity considerations into the functional analysis and allocation step. The framework's emphasis on physical and functional interface management for the different circularity principles and its guidance on allocating circularity functions appear to resonate with practitioners' experiences and needs while also recognizing the collaborative nature of this process.

6.2.4 Circularity Integrated Design Synthesis

The Design Synthesis step received the lowest average applicability rating of 3.6, revealing some challenges in integrating circularity principles into this phase of the SE process. The interviews uncovered a tension between the desire to influence design for circularity and the traditional role of systems engineers. **P3** pointedly asked:

"It says design for minimal material use, for example, but the system engineer is not responsible for the design."

This comment highlights a key challenge in implementing the framework to further consider systems engineer's influence on design decisions to facilitate circularity implementations without overstepping their conventional boundaries. An interesting perspective came from **P1**, who suggested a broader view of circularity in design:

" I think you should use the MKI as well. Because you do not want to be circular, despite of other environmental impacts."

This insight underscores the importance of considering the impact of circularity in the design trade studies in broader sense, rather than in isolation.

6.2.5 Circularity Integrated Verification and Validation

The Verification and Validation step received an average applicability rating of 4.2, indicating that interviewees saw this as an area where systems engineers could readily incorporate circularity principles. However, the discussions revealed some challenges in verifying and validating circularity requirements. **P1** highlighted a key difficulty:

"The most complicated part is to say it is recyclable at the end of the lifetime, but that is in 20 years. So how are you going to verify it now, especially when you have new products?"

This points to a fundamental challenge in circularity regarding how to verify long-term outcomes in the short term. $\mathbf{P2}$ offered an interesting perspective on the familiarity of the process:

"This is everyday work for me. Of course the topic is different, but the topic of every requirement is different. So it is just verifying and validating like an everyday job."

This suggests that while the content may be new, the process of verification and validation is a well-established practice for systems engineers, making the integration of circularity principles more straightforward to comprehend and implement.

6.2.6 Framework Validation Summary and Conclusion

Table 4 below summarizes the validation insights of the developed framework into strengths, potential improvements, and applicability score of actions for each step of the SE process. The validation process showed that the framework is effective and applicable, especially

Framework steps	Framework Strengths	Potential Improvements	Applicability (out of 5)	
Project Start	Clear starting point for integra- tion. Encourages consideration of all circularity principles.	Assign ownership of targets. Clarify the role of the systems engineer in setting targets.	4.2	
Requirements Analysis	Distinguishes specific and gen- eral circularity requirements. Encourages end-of-life considera- tions.	Involve technical experts. Provide more guidance on making circularity requirements SMART.	4	
Functional Anal- ysis & Allocation	Integrates circularity into func- tional thinking. Emphasizes interface manage- ment.	Provide more guidance on incor- porating circularity into function allocation. Include examples of circularity- influenced functional architec- ture.	4	
Design Synthesis	Encourages circularity in design decisions. Promotes documentation of dis- assembly plans.	Clarify systems engineer's role in design decisions. Provide strategies for balancing circularity with other design con- straints.	3.6	
Verification & Validation	Addresses long-term circularity goals. Distinguishes between verifica- tion and validation for circular- ity.	Develop new validation methods for long-term goals. Provide guidance on verifying circularity requirements.	4.2	
Note: Applicability score ranges from 1 to 5 for the actions in the framework, with 5 being highly applicable and 1 not applicable at all.				

Table 4: Strengths, potential improvements, and framework applicability.

in integrating circular economy principles during the project start and verification phases. Industry professionals from ProcessMinded confirmed its practical relevance but pointed out areas that need refinement, particularly in the design synthesis phase and the role of systems engineers. While the framework has strong potential, the validation suggests that further iteration and adaptation are needed to fully address the complexities of realworld application in construction projects. Overall, the framework is well-validated but requires targeted improvements for comprehensive implementation.

7 Discussion

This study explored how circular economy principles can be integrated into the systems engineering process within the construction industry. The research resulted in the development of a framework through literature review and expert validation aimed at guiding ProcessMinded and other practitioners in implementing this integration. This section interprets the key findings, compares these findings with existing literature, discusses scientific and practical contributions, and addresses limitations and future research directions.

7.1 Summary of Key Findings

This study has yielded several significant findings regarding the integration of circular economy principles into the SE process. The research demonstrates that incorporating circularity principles into each step of the SE process is applicable and potentially transformative. This finding aligns with the evolving nature of SE as a dynamic field capable of adapting to new challenges (Presland et al., 2018; M. Watson et al., 2022). The successful integration of circularity principles across all steps of the SE process suggests that SE is a suited comprehensive approach for incorporating circular economy principles in construction projects.

The study revealed that the project start step is crucial for setting the foundation for circularity throughout the project lifecycle. This finding underscores the importance of early-phase planning in circularity integration, as highlighted by Sanchez and Haas (2018). By incorporating circularity principles at the outset, projects can establish clear goals, align stakeholders, and create a shared vision for circularity outcomes. This early integration can help overcome traditional barriers to circularity implementation, such as lack of awareness or commitment among stakeholders, as identified by Adams et al. (2017)in their study on circularity challenges in construction. In the requirements analysis step, the framework's approach to distinguishing between specific and generic circularity requirements emerged as a key finding. This distinction offers a clear way to address circularity requirements, covering both broad requirements and specific measurable requirements. This aligns with the approach suggested by the European Commission (2009) in their ecodesign guidelines, which differentiate between generic and specific ecodesign requirements. However, the study also indicated ongoing challenges in translating broad circularity goals and targets into concrete requirements, reflecting the complexities of requirements engineering (Aravici et al., 2005; Kamara & Anumba, 2002; Wheeler, 2004). Furthermore, the functional analysis and allocation step showed strong potential for circularity integration, particularly in its emphasis on interface management and allocation of circularity functions. This step can extend traditional functional decomposition to include circularity aspects aligning with circular design strategies, especially the principles of designing for disassembly and adaptability (Eberhardt et al., 2022). This phase of SE can be leveraged to embed circularity thinking into the functional thinking of SE for construction projects, potentially leading to a more circular process. The design synthesis step emerged as the most challenging for circularity integration based on the expert's validation. This finding reflects a tension between the desire to influence design for circularity and the traditional role of systems engineers as process facilitators rather than design specialists. It highlights a potential need for redefining roles and responsibilities

within SE for circular construction projects. The study also uncovered challenges in verifying and validating long-term circularity outcomes. This finding underscores the need for new methods and models for verification and validation, as suggested by INCOSE (2021) in their 2035 vision for systems engineering. The long-term nature of many circularity requirements, such as end-of-life recyclability or adaptability over long periods, poses unique challenges for traditionally accepted verification and validation methods.

Although circularity principles can be integrated into all steps of the SE process, the degree of integration and the challenges encountered differ between steps. This indicates that the tailored, step-specific approach of the framework developed in this study is effective for integrating the four circular economy principles of closing, narrowing, slowing, and regenerating the resource loop. This makes the framework useful for organizations that apply all the steps of the SE process or those focused on specific steps within the process. This finding is consistent with R. S. de Graaf et al. (2017) and R. de Graaf et al. (2016), who noted that in some construction projects, only certain steps of SE process are applied. For the application of the framework, this is considered good as systems engineers can use the framework focusing on the processes they need or apply.

7.2 Research Contribution

This study makes several contributions to the scientific understanding of how circular economy principles are integrated into the SE process and to the practical application of these principles in the construction industry.

Firstly, this research addresses the gap identified by Gasparri et al. (2023) for sectorspecific frameworks and interdisciplinary approaches to circular economy implementation in construction. By developing a framework that integrates circular economy principles into the established SE process, this study provides a novel approach that bridges two previously distinct fields. This integration contributes to the broader body of knowledge in both SE and circular economy studies, offering a new theoretical perspective on how these concepts can be combined.

Secondly, the framework developed in this study responds to the call for more practical applications and frameworks for implementing circular economy principles in the construction industry, as highlighted by Antwi-Afari et al. (2021) and Guerra et al. (2021). By providing a structured approach for integrating circularity into each step of the SE process, this research offers a framework that can be applied and further studied in real-world construction projects. This addresses the critical need for actionable methodologies to translate circular economy theory into practice within the construction industry.

Thirdly, this study contributes to addressing the demand for new design typologies and decision-making guides for circular economy implementation, as identified by Eberhardt et al. (2022). The framework's approach to integrating circularity considerations into the functional analysis and allocation, and design synthesis steps provides a new perspective on how circular design principles can be systematically incorporated into the design process. This contribution has the potential to influence how circular design is conceptualized and implemented in construction projects.

Furthermore, for the practical contribution, this research offers considerable value to ProcessMinded and similar organizations in the construction industry. The developed framework provides ProcessMinded with a structured approach to incorporating circularity principles into their existing SE practices. This addresses ProcessMinded's specific need for a widely applicable approach to incorporating circularity into SE that is generic enough to be used across different projects yet flexible enough to incorporate different circular economy principles. This contribution has the potential to enhance the current practices of organizations toward circularity in the construction industry. In addition, the validation of the framework through interviews with industry professionals provides initial evidence of its applicability and relevance to real-world practices. This not only contributes to the practical validation of the framework but also highlights areas for future refinement and research, paying the way for further studies in this field. By focusing on the integration of circular economy principles into SE, this research contributes to the ongoing evolution of SE as a discipline, aligning with the vision set out by INCOSE (2021) for the future of SE, which emphasizes the need for evolution to address complex interconnected challenges.

7.3 Limitations and Future Research

While this research contributes valuable insights, limitations exist that should be acknowledged, providing a basis for future research. These limitations are discussed below, then suggestion for future research are provided.

7.3.1 Limitations

The primary limitation arises from the framework validation process, which relied on interviews with only five professionals from ProcessMinded. This small sample size, confined to a single company, limits the generalizability of the findings and may not fully represent the diverse perspectives within the broader construction industry. The validation focused heavily on the actions in the framework within the different SE steps. While informative for the study, this may not capture the full complexity of integrating circularity principles in real-world projects. In addition, the iterative nature of SE processes and the different levels of detail are not sufficiently addressed in the framework presented in this study, which could make differences in the framework structure and expand it further or imply changes. Another noteworthy limitation is the framework's theoretical nature. Although validated by experts, the framework has not been tested in actual construction projects. This lack of practical implementation means that unforeseen challenges or limitations in applying the framework may exist that were not captured in this study.

Additionally, the research also focused primarily on the technical aspects of SE, specifically the technical SE process, potentially overlooking important organizational and management processes that could influence the integration of circular economy principles. The framework's emphasis on the SE process and ProcessMinded's ten-step approach to systems engineering may limit its applicability to organizations with different SE methodologies. Also, the study's approach to circularity, focusing on four main principles (closing, narrowing, slowing, and regenerating the resource loop), while comprehensive, may not capture all nuances of circular economy concepts for and within the construction industry. The integration of these principles into each SE step was based on theoretical connections based on literature, which may oversimplify the complexities of real-world application. An additional limitation to consider is the potential for misinterpretation of context-specific terms with the integration. The study draws from diverse fields such as but not limited to systems engineering, circular economy, construction management, product manufacturing, stakeholders management, and requirements engineering. While efforts were made to accurately interpret and synthesize literature, there is a possibility that some context-specific meanings may have been misunderstood. This is particularly relevant when applying concepts from one field to another, where terminologies and methodologies may not always align perfectly. For instance, the interpretation of circular economy principles and their application to systems engineering steps relied on the researcher's understanding and synthesis of various sources. While care was taken to maintain accuracy, the complexity of both fields leaves room for potential misinterpretations that could affect the framework's theoretical foundation. Lastly, the research did not extensively address the potential conflicts between different circularity principles. This limitation was highlighted in the validation interviews, where tensions between different circularity aspects (e.g., durability vs. recyclability) were noted but not fully explored within the framework.

7.3.2 Future Research

Building upon these limitations, the following future research directions are recommended to further build on this study:

- Diverse Sample Validation: Future research could validate the framework with a larger and more diverse sample of professionals across various organizations in the construction industry. This broader validation would enhance the framework's generalizability and provide a more comprehensive understanding of its applicability.
- The Iterative nature of SE and different levels of details: Further investigation of the iterative nature of SE and how the framework could be applied at different levels of detail can potentially reveal valuable insights. Additionally, the actions could be extended to include the requirements analysis loop and design loop reflecting what the integration would require when revisiting previous steps throughout the whole process.
- Real-World Application: To move beyond theoretical validation, future research could implement the framework in actual construction projects. Longitudinal studies could provide valuable insights into the practical application of circular economy principles at each step of the SE process. This could help explore and identify real-world challenges and insights.
- Exploration of Organizational and Management Aspects of SE: Future research could investigate how different organizational structures and management processes within SE affect the integration of circular economy principles. This includes exploring potential differences between client and contractor organizations when applying SE methods and how these variations influence the framework's adoption and effectiveness.
- Addressing Conflicts in Circularity Principles: Future research could study further the conflicts between various circularity principles within the SE process. For example, exploring the trade-offs between different circular strategies streaming from

different principles can provide systems engineers with practical insights for balancing these aspects in project implementation.

By addressing these limitations and pursuing these suggested research directions, the academic and professional communities can build on this study's foundation. This continued exploration could refine the integration of circular economy principles into SE practices and eventually contribute to more sustainable and circular practices within the construction industry.

8 Conclusion

This research set out to explore the integration of circular economy principles into the SE process for the construction industry, with a specific focus on developing a framework for ProcessMinded to implement this integration. The study has successfully addressed its main research question and associated sub-questions, providing valuable insights into how circularity can be integrated within SE.

First, the study provided a comprehensive understanding of the systems engineering processes and circular economy principles relevant to the construction industry. It established the foundational concepts necessary for integrating the four key circular economy principles: closing, narrowing, slowing, and regenerating the resource loop with the five SE processes: project start, requirements analysis, functional analysis & allocation, design synthesis, and verification & validation.

Second, a structured framework was developed, presented in the form of inputs, actions, and outputs for each SE process to guide the integration of these four principles across each step of the SE process. The framework offers a systematic approach, detailing specific actions and expected outputs for systems engineers. Validation through expert interviews confirmed the framework's applicability, with particular strength in the project start and verification steps. However, challenges were noted in the design synthesis step, where the integration of circularity proved more complex and technically demanding, highlighting areas for further development. This variation in applicability based on systems engineers' assessments suggests that they may find the integration of circularity principles more straightforward in certain steps that they are familiar with and more technically challenging in others, particularly when specific technical expertise is needed.

Third, the research delivered a practical, actionable framework that ProcessMinded and similar organizations can use to integrate circular economy principles into their SE practices. The inclusion of practical examples for some actions enhances the framework's relevance and utility, addressing the pressing need for more circular approaches in the construction industry.

In summary, while the study has its limitations, particularly concerning the scope of validation and its theoretical nature, it lays a solid foundation for future research. The identified areas for further investigation, including practical implementation, offer promising research directions for further studies in this field. This research not only advances the theoretical understanding of integrating circularity with the SE process but also provides practical applications for ProcessMinded, serving as a stepping stone toward more circular practices through SE in the construction industry.

References

- Adams, K. T., Osmani, M., Thorpe, T., & Thornback, J. (2017). Circular economy in construction: Current awareness, challenges and enablers. Proceedings of Institution of Civil Engineers: Waste and Resource Management, 170(1), 15–24. https: //doi.org/10.1680/jwarm.16.00011
- Ahmed, S., Majava, J., & Aaltonen, K. (2023). Implementation of circular economy in construction projects: a procurement strategy approach. *Construction Innovation*, 24(7), 204–222. https://doi.org/10.1108/CI-12-2022-0327/FULL/PDF
- Akinade, O. O., Oyedele, L. O., Ajayi, S. O., Bilal, M., Alaka, H. A., Owolabi, H. A., Bello, S. A., Jaiyeoba, B. E., & Kadiri, K. O. (2017). Design for Deconstruction (DfD): Critical success factors for diverting end-of-life waste from landfills. Waste Management, 60, 3–13. https://doi.org/10.1016/J.WASMAN.2016.08.017
- Akinade, O. O., Oyedele, L. O., Omoteso, K., Ajayi, S. O., Bilal, M., Owolabi, H. A., Alaka, H. A., Ayris, L., & Henry Looney, J. (2017). BIM-based deconstruction tool: Towards essential functionalities. *International Journal of Sustainable Built Environment*, 6(1), 260–271. https://doi.org/10.1016/j.ijsbe.2017.01.002
- Al-Jibouri, S., & Ogink, G. (2009). Proposed model for integrating RAMS method in the design process in construction. Architectural Engineering and Design Management, 5(4), 179–192. https://doi.org/10.3763/aedm.2008.0100
- Allwood, J. M., Ashby, M. F., Gutowski, T. G., & Worrell, E. (2011). Material efficiency: A white paper. *Resources, Conservation and Recycling*, 55(3), 362–381. https://doi.org/10.1016/J.RESCONREC.2010.11.002
- Anastasiades, K., Blom, J., & Audenaert, A. (2023). Circular Construction Indicator: Assessing Circularity in the Design, Construction, and End-of-Life Phase. *Recycling 2023*, Vol. 8, Page 29, 8(2), 29. https://doi.org/10.3390/RECYCLING8020029
- Andrich, W., Daniotti, B., Pavan, A., & Mirarchi, C. (2022). Check and Validation of Building Information Models in Detailed Design Phase: A Check Flow to Pave the Way for BIM Based Renovation and Construction Processes. *Buildings 2022, Vol.* 12, Page 154, 12(2), 154. https://doi.org/10.3390/BUILDINGS12020154
- Antwi-Afari, P., Ng, S. T., Chen, J., Oluleye, B. I., Antwi-Afari, M. F., & Ababio, B. K. (2023). Enhancing life cycle assessment for circular economy measurement of different case scenarios of modular steel slab. *Building and Environment*, 239, 110411. https://doi.org/10.1016/J.BUILDENV.2023.110411
- Antwi-Afari, P., Ng, S. T., & Hossain, M. U. (2021). A review of the circularity gap in the construction industry through scientometric analysis. *Journal of Cleaner Production*, 298. https://doi.org/10.1016/j.jclepro.2021.126870
- Arayici, Y., Aouad, G., & Ahmed, V. (2005). Requirements engineering for innovative integrated ICT systems for the construction industry. *Construction Innovation*, 5(3), 179–200. https://doi.org/10.1191/1471417505ci097oa
- ArchDaily. (2016). The Edge / PLP Architecture ArchDaily. https://www.archdaily. com/785967/the-edge-plp-architecture
- Attia, S. (2018). Regenerative and Positive Impact Architecture Roadmap, 81–94. https://doi.org/10.1007/978-3-319-66718-8{_}8
- Azhar, S., Khalfan, M., & Maqsood, T. (2012). Building information modeling (BIM): Now and beyond. Australasian Journal of Construction Economics and Building, 12(4), 15–28. https://doi.org/10.5130/ajceb.v12i4.3032
- B. Pena, R. (2014). Living Proof THE BULLITT CENTER.

- Badampudi, D., Wohlin, C., & Petersen, K. (2015). Experiences from using snowballing and database searches in systematic literature studies. ACM International Conference Proceeding Series, 27-29-April-2015. https://doi.org/10.1145/2745802. 2745818
- Bal, M., Bryde, D., Fearon, D., & Ochieng, E. (2013). Stakeholder Engagement: Achieving Sustainability in the Construction Sector. Sustainability (Switzerland), 5(2), 695– 710. https://doi.org/10.3390/su5020695
- Baldwin, E. (2021). "Let's Change the Culture of Construction": Creating New Ways of Building with Superuse Studios — ArchDaily. https://www.archdaily.com/ 958609/lets-change-the-culture-of-construction-creating-new-ways-of-buildingwith-superuse-studios
- Baumeister, R. F., & Leary, M. R. (1997). Writing Narrative Literature Reviews. https://doi.org/10.103' 2680.1.3.311, 1(3), 311–320. https://doi.org/10.1037/1089-2680.1.3.311
- Blanchard, B. S., & Fabrycky, W. J. (2005). Systems Engineering and Analysis, fourth edition. *International Series in Industrial and Systems Engineering*, (March), 786.
- Blok, M. (2024). From Buildings to Resources: The Role of Urban Mining in Circular Cities. https://www.metabolic.nl/news/urban-mining-and-circular-construction/
- Blyth, A., & Worthington, J. (2002). Managing the Brief for Better Design. Managing the Brief for Better Design. https://doi.org/10.4324/9780203478349
- Bocken, N. M., Farracho, M., Bosworth, R., & Kemp, R. (2014). The front-end of ecoinnovation for eco-innovative small and medium sized companies. *Journal of En*gineering and Technology Management - JET-M, 31(1), 43–57. https://doi.org/ 10.1016/J.JENGTECMAN.2013.10.004
- Bocken, N., Stahel, W., Dobrauz, G., Koumbarakis, A., Obst, M., & Matzdorf, P. (2021). Circularity as the new normal: Future fitting Swiss business strategies. https: //cris.maastrichtuniversity.nl/en/publications/circularity-as-the-new-normalfuture-fitting-swiss-business-strat
- Bocken, N. M., de Pauw, I., Bakker, C., & van der Grinten, B. (2016). Product design and business model strategies for a circular economy. *Journal of Industrial and Production Engineering*, 33(5), 308–320. https://doi.org/10.1080/21681015.2016. 1172124
- Boulding, K. E. (1966). The Economics of the Coming Spaceship Earth. In H. Jarrett (Ed.), Environmental Quality in a Growing Economy (pp. 3-14). Baltimore, MD: Resources for the Future/Johns Hopkins University Press.
- Bouyarmane, H., El Amine, M., & Sallaou, M. (2020). Disassembly evaluation during the conceptual design phase to ensure a better valorisation of products at the end of life cycle. *International Journal of Sustainable Development and Planning*, 15(5), 673–684. https://doi.org/10.18280/IJSDP.150510
- Bragança, L. (2019). SBE19 Brussels BAMB-CIRCPATH "Buildings as Material Banks
 A Pathway For A Circular Future". *IOP Conference Series: Earth and Environmental Science*, 225(1), 011001. https://doi.org/10.1088/1755-1315/225/1/011001
- Bragança, L., Vieira, S. M., & Andrade, J. B. (2014). Early stage design decisions: The way to achieve sustainable buildings at lower costs. *The Scientific World Journal*, 2014. https://doi.org/10.1155/2014/365364
- Brand, S. (1995). How Buildings Learn: What Happens After They're Built.
- BRE. (n.d.). Building Research Establishment: Testing, Inspection and Certification services. https://bregroup.com/services/testing-inspection-certification

- Brook, P., Madni, A. M., Pennotti, M., Rousseau, D., & Sillito, H. (2024). Five Perspectives on Transdisciplinary Systems Engineering. *INSIGHT*, 27(2), 21–26. https://doi.org/10.1002/INST.12483
- Buck, P., Dronkers, J. H., van der Linde, P., Nijpels, E., Siertsema, T., & Verhagen, M. (2023). Leidraad voor Systems Engineering binnen de GWW-sector: General description on application of SE in the Dutch infrastructure construction sector (tech. rep.). https://www.leidraadse.nl/
- Buede, D. M. (2000). The engineering design of systems : models and methods, 462.
- Buildings. (2012). Empire State Building Beats Energy Savings Goal Buildings. https: //www.buildings.com/home/article/10189556/empire-state-building-beatsenergy-savings-goal
- Carbon Leadership Forum. (2023). Embodied Carbon in Construction Calculator (EC3) - Carbon Leadership Forum. https://carbonleadershipforum.org/ec3-tool/
- Çetin, S., De Wolf, C., & Bocken, N. (2021). Circular digital built environment: An emerging framework. Sustainability (Switzerland), 13(11). https://doi.org/10. 3390/su13116348
- Cheshire, D. (2016). Building revolutions : applying the circular economy to the built environment, 138. https://www.cip-books.com/product/building-revolutionsapplying-the-circular-economy-to-the-built-environment/
- CIEEM. (2017). Guidelines for Preliminary Ecological Appraisal Second Edition. www. cieem.net
- Çimen, Ö. (2021). Construction and built environment in circular economy: A comprehensive literature review. Journal of Cleaner Production, 305, 127180. https:// doi.org/10.1016/J.JCLEPRO.2021.127180
- Coenen, T. B., Haanstra, W., Jan Braaksma, A. J., & Santos, J. (2020). CEIMA: A framework for identifying critical interfaces between the Circular Economy and stakeholders in the lifecycle of infrastructure assets. *Resources, Conservation and Recycling*, 155, 104552. https://doi.org/10.1016/J.RESCONREC.2019.104552
- Cornet, S. M., den Berg, v. M., & Oorschot JAWH, v. (2016). D1 Synthesis of the stateof-the-art Key barriers and opportunities for Materials Passports and Reversible Building Design in the current system.
- Dams, B., Maskell, D., Shea, A., Allen, S., Driesser, M., Kretschmann, T., Walker, P., & Emmitt, S. (2021). A circular construction evaluation framework to promote designing for disassembly and adaptability. *Journal of Cleaner Production*, 316, 128122. https://doi.org/10.1016/J.JCLEPRO.2021.128122
- David, I., Bork, D., & Kappel, G. (2024). Circular systems engineering. Software and Systems Modeling 2024 23:2, 23(2), 269–283. https://doi.org/10.1007/S10270-024-01154-4
- de Bruijn, H., & ten Heuvelhof, E. (2018). Management in networks, second edition. Management in Networks, Second Edition, 1–149. https://doi.org/10.4324/ 9781315453019
- de Graaf, R. S., Vromen, R. M., & Boes, J. ((2017). Applying systems engineering in the civil engineering industry: an analysis of systems engineering projects of a Dutch water board. *Civil Engineering and Environmental Systems*, 34(2), 144– 161. https://doi.org/10.1080/10286608.2017.1362399
- de Graaf, R. (2020). Handbook Systems Engineering in the construction industry.

- de Graaf, R., Voordijk, H., & Van Den Heuvel, L. (2016). Implementing Systems Engineering in the Civil Engineering Consulting Firm: An Evaluation. Systems Engineering, 19(1), 44–58. https://doi.org/10.1002/sys.21336
- Debska, B., Caetano, M. A., & Brigolini Silva, G. J. (2024). Study of the influence of accelerated aging on the physical and mechanical properties of polymer composites containing rubber, polyethylene and poly(ethylene terephthalate) waste. *Journal* of Cleaner Production, 444, 141273. https://doi.org/10.1016/J.JCLEPRO.2024. 141273
- Desai, A., & Mital, A. (2003). Evaluation of disassemblability to enable design for disassembly in mass production. *International Journal of Industrial Ergonomics*, 32(4), 265–281. https://doi.org/10.1016/S0169-8141(03)00067-2
- Dewagoda, K. G., Thomas Ng, S., & Kumaraswamy, M. M. (2022). Design for Circularity: The Case of the Building Construction Industry. *IOP Conference Series: Earth* and Environmental Science, 1101(6), 062026. https://doi.org/10.1088/1755-1315/1101/6/062026
- DGNB. (n.d.). DGNB Certification for Buildings DGNB. https://www.dgnb.de/en/ certification/buildings
- DiCicco-Bloom, B., & Crabtree, B. F. (2006, April). The qualitative research interview. https://doi.org/10.1111/j.1365-2929.2006.02418.x
- Dove, R., & LaBarge, R. (2014). 8.4.2 Fundamentals of Agile Systems Engineering Part 2. INCOSE International Symposium, 24(1), 876–892. https://doi.org/10.1002/ J.2334-5837.2014.TB03187.X
- Durmisevic, E., Jorissen, A., Eindhoven, T. U., Nijsen, R., Delft, T. U., Broek, K., Jacobs, J., Stoutjesdijk, P., Ribberink, M., & Nederand, P. (2019). CIRCULAR ECONOMY IN CONSTRUCTION DESIGN STRATEGIES FOR REVERSIBLE BUILDINGS.
- Dykes, K., Meadows, R., Graf, P., Hand, M., Lunacek, M., Michalakes, J., Moriarty, P., Musial, W., & Veers, P. (2011). Applications of Systems Engineering to the Research, Design, and Development of Wind Energy Systems With contributions from: F. http://www.osti.gov/bridge
- Eberhardt, L. C. M., Birkved, M., & Birgisdottir, H. (2022). Building design and construction strategies for a circular economy. Architectural Engineering and Design Management, 18(2), 93–113. https://doi.org/10.1080/17452007.2020.1781588
- Ecochain. (2024). ESPR Guide What is it & How to prepare (2024 Update) Ecochain. https://ecochain.com/blog/espr-2024-overview/
- Ellen MacArthur Foundation. (2013). Founding Partners of the Ellen MacArthur Foundation 2013 CIRCULAR ECONOMY TOWARDS THE Economic and business rationale for an accelerated transition (tech. rep.).
- Ellen Macarthur Foundation. (2015). Towards a circular economy: business rationale for an accelerated transition. https://www.ellenmacarthurfoundation.org/towards-acircular-economy-business-rationale-for-an-accelerated-transition
- Ellen Macarthur Foundation. (2022). Why buy light bulbs when you can buy light? Signify. https://www.ellenmacarthurfoundation.org/circular-examples/why-buy-light-bulbs-when-you-can-buy-light-signify
- $\label{eq:EUR-Lex.} \begin{array}{l} \text{EUR-Lex. (2022). EUR-Lex 52022XC0504(01) EN EUR-Lex. https://eur-lex.europa.} \\ \text{eu/legal-content/EN/TXT/?uri=uriserv:OJ.C_.2022.182.01.0001.01.ENG} \end{array}$
- European Commission. (2009). DIRECTIVE 2009/125/EC OF THE EUROPEAN PAR-LIAMENT AND OF THE COUNCIL of 21 October 2009 - establishing a frame-

work for the setting of ecodesign requirements for energy-related products. https://eur-lex.europa.eu/legal-content/EN/ALL/?uri=celex%3A32009L0125

- European Energy Innovation. (2017). BAMB Buildings As Material Banks: Where are we now? https://www.europeanenergyinnovation.eu/Latest-Research/Summer-2017/BAMB-Buildings-As-Material-Banks-Where-are-we-now
- Eurostat. (2023). Eurostat, Environment and energy. https://ec.europa.eu/eurostat/ databrowser/view/env_wasgen/default/table?lang=en
- Ferrari, R. (2015). Writing narrative style literature reviews. *Medical Writing*, 24(4), 230–235. https://doi.org/10.1179/2047480615Z.00000000329
- Ferris, T. L. (2007a). 7.4.3 Some Early History of Systems Engineering 1950's in IRE Publications (Part 1): The Problem. *INCOSE International Symposium*, 17(1), 1080–1095. https://doi.org/10.1002/J.2334-5837.2007.TB02935.X
- Ferris, T. L. (2007b). 7.4.4 Some Early History of Systems Engineering 1950's in IRE Publications (Part 2): The Solution. INCOSE International Symposium, 17(1), 1096–1111. https://doi.org/10.1002/J.2334-5837.2007.TB02936.X
- Fobbe, L., & Hilletofth, P. (2023). Moving toward a circular economy in manufacturing organizations: the role of circular stakeholder engagement practices. *International Journal of Logistics Management*, 34 (3), 674–698. https://doi.org/10.1108/IJLM-03-2022-0143
- Forsberg, K., & Mooz, H. (1992). The Relationship of Systems Engineering to the Project Cycle. Engineering Management Journal, 4(3), 36–43. https://doi.org/10.1080/ 10429247.1992.11414684
- Gálvez-Martos, J. L., Styles, D., Schoenberger, H., & Zeschmar-Lahl, B. (2018). Construction and demolition waste best management practice in Europe. *Resources, Conservation and Recycling*, 136, 166–178. https://doi.org/10.1016/j.resconrec. 2018.04.016
- Gasparri, E., Arasteh, S., Kuru, A., Stracchi, P., & Brambilla, A. (2023). Circular economy in construction: A systematic review of knowledge gaps towards a novel research framework. https://doi.org/10.3389/fbuil.2023.1239757
- Gerding, D. P., Wamelink, H., & Leclercq, E. M. (2021). Implementing circularity in the construction process: a case study examining the reorganization of multi-actor environment and the decision-making process. *Construction Management and Economics*, 39(7), 617–635. https://doi.org/10.1080/01446193.2021.1934885
- González, A., Sendra, C., Herena, A., Rosquillas, M., & Vaz, D. (2021). Methodology to assess the circularity in building construction and refurbishment activities. *Re*sources, Conservation and Recycling Advances, 12. https://doi.org/10.1016/j. rcradv.2021.200051
- Göswein, V., Carvalho, S., Cerqueira, C., & Lorena, A. (2022). Circular material passports for buildings - Providing a robust methodology for promoting circular buildings. *IOP Conference Series: Earth and Environmental Science*, 1122(1). https://doi. org/10.1088/1755-1315/1122/1/012049
- Government of the Netherlands. (2019). Climate Agreement (tech. rep.). https://www.klimaatakkoord.nl/documenten/publicaties/2019/06/28/national-climate-agreement-the-netherlands
- GOV.UK. (2022). HS2 Biodiversity Accounting Reports. https://www.gov.uk/government/ publications/hs2-biodiversity-accounting-reports

- Greenhalgh, T., Thorne, S., & Malterud, K. (2018). Time to challenge the spurious hierarchy of systematic over narrative reviews? *European Journal of Clinical Investi*gation, 48(6), e12931. https://doi.org/10.1111/ECI.12931
- Guerra, B. C., Shahi, S., Molleai, A., Skaf, N., Weber, O., Leite, F., & Haas, C. (2021). Circular economy applications in the construction industry: A global scan of trends and opportunities. *Journal of Cleaner Production*, 324. https://doi.org/10.1016/ j.jclepro.2021.129125
- Guy, B., & Ciarimboli, N. (2005). DfD Design for Disassembly in the built environment: a guide to closed-loop design and building Foreword and Acknowledgements (tech. rep.).
- Heinrich, M., & Lang, W. (2019). MATERIALS PASSPORTS-BEST PRACTICE. https: //www.bamb2020.eu/wp-content/uploads/2019/02/BAMB_MaterialsPassports_ BestPractice.pdf
- Heisel, F., & Rau-Oberhuber, S. (2020). Calculation and evaluation of circularity indicators for the built environment using the case studies of UMAR and Madaster. *Journal of Cleaner Production*, 243, 118482. https://doi.org/10.1016/J.JCLEPRO. 2019.118482
- Henderson, Z., Morgan, B., & Watson, J. (2019). Implementing the circular economy in buildings: Processes of sensemaking in design and delivery. Association of Researchers in Construction Management, ARCOM 2019 - Proceedings of the 35th Annual Conference, 114–123.
- HM Government. (2013). Construction 2025: industrial strategy for construction government and industry in partnership. https://www.gov.uk/government/publications/ construction-2025-strategy
- HM Treasury. (2013). Infrastructure Carbon Review (tech. rep.).
- Hoehne, O. (2023). Lessons Learned and Recommendations for the Application of Systems Engineering as an Emerging Discipline in Transportation & Infrastructure Projects. *INCOSE International Symposium*, 33(1), 48–70. https://doi.org/10. 1002/IIS2.13008
- Hoehne, O. M. (2012). 5.1.2 Entering a Brave New World: Applying Systems Engineering to American Infrastructure Projects. *INCOSE International Symposium*, 22(1), 663–677. https://doi.org/10.1002/J.2334-5837.2012.TB01363.X
- Hossain, M. U., Ng, S. T., Antwi-Afari, P., & Amor, B. (2020, September). Circular economy and the construction industry: Existing trends, challenges and prospective framework for sustainable construction. https://doi.org/10.1016/j.rser.2020. 109948
- Hossain, N. U. I., Jaradat, R. M., Hamilton, M. A., Keating, C. B., & Goerger, S. R. (2020). A Historical Perspective on Development of Systems Engineering Discipline: A Review and Analysis. *Journal of Systems Science and Systems Engineering*, 29(1), 1–35. https://doi.org/10.1007/S11518-019-5440-X/METRICS
- Hubmann, G., & van Maaren, V. (2022). Circular Material Systems: anticipating wholesystem design in architecture and construction. IOP Conference Series: Earth and Environmental Science, 1078(1). https://doi.org/10.1088/1755-1315/1078/1/ 012002
- Incelli, F., Cardellicchio, L., & Rossetti, M. (2023). Circularity Indicators as a Design Tool for Design and Construction Strategies in Architecture. *Buildings*, 13(7). https://doi.org/10.3390/buildings13071706

- Inchainge. (n.d.). How can we measure Circularity in a Circular Economy? https://inchainge.com/resources/blogs/how-can-we-measure-circularity/
- INCOSE. (2015). Systems Engineering Handbook: A Guide for System Life Cycle Processes and Activities (4th).
- INCOSE. (2021). Systems Engineering Vision 2035. www.incose.org/sevision
- Indwar, R., & Titiksh, A. (2024). Organic wastes as a sustainable alternative to synthetic thermal insulating materials. *Environment, Development and Sustainability*, 1–41. https://doi.org/10.1007/S10668-024-05207-Y/TABLES/1
- International Monetary Fund. European Dept. (2023). Housing Supply in The Netherlands: The Road to More Affordable Living 1 (tech. rep.). https://wetten.overheid. nl/BWBR0003237/2022-07-01.
- ISO. (2022). Circular economy The role of conformity assessment. International Organization for Standardization. https://www.iso.org/publication/PUB100471.html
- ISO/IEC/IEEE 15288. (2015). Systems and software engineering System life cycle processes. https://www.iso.org/standard/63711.html
- Iyer-Raniga, U. (2019). Using the ReSOLVE framework for circularity in the building and construction industry in emerging markets. IOP Conference Series: Earth and Environmental Science, 294(1). https://doi.org/10.1088/1755-1315/294/1/ 012002
- Jensen, P. A. (2011). Inclusive briefing and user involvement: Case study of a media centre in denmark. Architectural Engineering and Design Management, 7(1), 38– 49. https://doi.org/10.3763/aedm.2010.0124
- Kamara, J. M., & Anumba, C. J. (2002). Requirements Capture in Building Design. The Design Journal, 5(1), 50–61. https://doi.org/10.2752/146069202790718558
- Kibert, c. J. (2016). Green Building Design and Delivery. Sustainable Construction, 1(4), 1–36. https://books.google.com/books/about/Sustainable_Construction.html? id=2xgWCgAAQBAJ
- Kirchherr, J., Reike, D., & Hekkert, M. (2017). Conceptualizing the circular economy: An analysis of 114 definitions. https://doi.org/10.1016/j.resconrec.2017.09.005
- Konietzko, J., Bocken, N., & Hultink, E. J. (2020). Circular ecosystem innovation: An initial set of principles. Journal of Cleaner Production, 253. https://doi.org/10. 1016/J.JCLEPRO.2019.119942
- Kossiakoff, A., Sweet, W. N., Seymour, S. J., & Biemer, S. M. (2011). Systems Engineering Principles and Practice: Second Edition. Systems Engineering Principles and Practice: Second Edition. https://doi.org/10.1002/9781118001028
- Kraaijenhagen, C., Oppen, C. v., Bocken, N., & Bernasco, C. (2016). Circular business : collaborate and circulate. https://www.copper8.com/en/en-book-test/
- Kujala, J., Sachs, S., Leinonen, H., Heikkinen, A., & Laude, D. (2022). Stakeholder Engagement: Past, Present, and Future. Business and Society, 61(5), 1136–1196. https://doi.org/10.1177/00076503211066595/ASSET/IMAGES/LARGE/10. 1177{_}00076503211066595-FIG2.JPEG
- Lehtinen, J., Kier, C., Aaltonen, K., & Huemann, M. (2023, January). A complexity perspective on project stakeholder management. In *Research handbook on complex* project organizing (pp. 243–253). Edward Elgar Publishing Ltd. https://doi.org/ 10.4337/9781800880283.00035
- Leidraad SE. (2007). Conference Presentation Guidelines SE. https://www.leidraadse. nl/nieuwsitem/78/Congres-Presentatie-Leidraad-SE

- Liu, Z., Osmani, M., Demian, P., & Baldwin, A. (2015). A BIM-aided construction waste minimisation framework. Automation in Construction, 59, 1–23. https://doi.org/ 10.1016/J.AUTCON.2015.07.020
- Living Future. (n.d.). Living Building Challenge Basics International Living Future Institute. https://living-future.org/lbc/basics4-0/
- Locatelli, G., Mancini, M., & Romano, E. (2014). Systems Engineering to improve the governance in complex project environments. *International Journal of Project Man*agement, 32(8), 1395–1410. https://doi.org/10.1016/J.IJPROMAN.2013.10.007
- Lynghaug, T. F., Kokkula, S., & Muller, G. (2022). Investigating Systems Engineering Approaches in the Norwegian Construction Industry: A Multi-Case Study. *INCOSE International Symposium*, 32(1), 792–808. https://doi.org/10.1002/iis2.12964
- Madaster. (2022). EPEA and Madaster simplify measuring circularity within the built environment - Madaster Global. https://madaster.com/inspiration/epea-andmadaster-simplify-measuring-circularity-within-the-built-environment/
- Madni, A. M. (2015). Expanding Stakeholder Participation in Upfront System Engineering through Storytelling in Virtual Worlds. Systems Engineering, 18(1), 16–27. https://doi.org/10.1002/SYS.21284
- Maes, J., & Jacobs, S. (2017). Nature-Based Solutions for Europe's Sustainable Development. Conservation Letters, 10(1), 121–124. https://doi.org/10.1111/CONL. 12216
- MannionMike & KeepenceBarry. (1995). SMART requirements. ACM SIGSOFT Software Engineering Notes, 20(2), 42–47. https://doi.org/10.1145/224155.224157
- Maylor, H., & Turner, N. (2017). Understand, reduce, respond: project complexity management theory and practice. International Journal of Operations and Production Management, 37(8), 1076–1093. https://doi.org/10.1108/IJOPM-05-2016-0263/FULL/PDF
- Mesmer, B., Mckinney, D., Watson, M., Madni, A. M., Mesmer, B., Mckinney, D., Martin, L., Watson, M., & Madni, A. M. (2022). Transdisciplinary Systems Engineering Approaches. *Recent Trends and Advances in Model Based Systems Engineering*, 579–590. https://doi.org/10.1007/978-3-030-82083-1{_}49
- Minunno, R., O'Grady, T., Morrison, G. M., & Gruner, R. L. (2020). Exploring environmental benefits of reuse and recycle practices: A circular economy case study of a modular building. *Resources, Conservation and Recycling*, 160, 104855. https: //doi.org/10.1016/J.RESCONREC.2020.104855
- Moraga, G., Huysveld, S., Mathieux, F., Blengini, G. A., Alaerts, L., Van Acker, K., de Meester, S., & Dewulf, J. (2019). Circular economy indicators: What do they measure? *Resources, Conservation and Recycling*, 146, 452–461. https://doi.org/ 10.1016/J.RESCONREC.2019.03.045
- Mostashari, A., McComb, S. A., Kennedy, D. M., Cloutier, R., & Korfiatis, P. (2012). Developing a stakeholder-assisted agile CONOPS development process. Systems Engineering, 15(1), 1–13. https://doi.org/10.1002/SYS.20190
- New Horizon. (n.d.). New Horizon New Horizon. https://newhorizon.nl/
- Oberholzer, S., & Sachs, S. (2023). Engaging stakeholders in the circular economy: A systematic literature review. Stakeholder Engagement in a Sustainable Circular Economy: Theoretical and Practical Perspectives, 57–97. https://doi.org/10.1007/ 978-3-031-31937-2{_}3

- Osaily, Y., Copping, A., & McCann, S. (2022). Demolition project management pathways: The route towards circular demolition. *IOP Conference Series: Earth and Envi*ronmental Science, 1101(6). https://doi.org/10.1088/1755-1315/1101/6/062002
- Osaily, Y., Copping, A., McCann, S., & Uddin, T. (2019). Exploring the value of demolition contractor involvement at the design stage of construction. Association of Researchers in Construction Management, ARCOM 2019 - Proceedings of the 35th Annual Conference, 334–343.
- Pacheco-Torgal, F., & Labrincha, J. A. (2013). Biotechnologies and bioinspired materials for the construction industry: an overview. https://doi.org/10.1080/19397038. 2013.844741
- Pelsmakers, S., Donovan, E., Hoggard, A., & Kozminska, U. (2022). Designing for the Climate Emergency : A Guide for Architecture Students. *Designing for the Climate Emergency*. https://doi.org/10.4324/9781003297246
- Peng, Y., Wang, D., Chu, D., Kladovasilakis, N., Pemas, S., & Pechlivani, E. M. (2024). Computer-Aided Design of 3D-Printed Clay-Based Composite Mortars Reinforced with Bioinspired Lattice Structures. *Biomimetics 2024, Vol. 9, Page 424, 9*(7), 424. https://doi.org/10.3390/BIOMIMETICS9070424
- Philips. (2015). Closing the materials loop Towards a circular economy Collection & Recycling.
- Phillips, R. A., & Reichart, J. (2000). The environment as a stakeholder? A fairness-based approach. *Journal of Business Ethics*, 23(2), 185–197. https://doi.org/10.1023/A: 1006041929249/METRICS
- Platform CB'23. (2022). Measuring Circularity Guide A measurement method for circular construction.
- Platform CB'23. (2023). Guide to Circular Design 2.0 Working agreements for circular construction.
- Pomponi, F., & Moncaster, A. (2017). Circular economy for the built environment: A research framework. Journal of Cleaner Production, 143, 710–718. https://doi. org/10.1016/J.JCLEPRO.2016.12.055
- Presland, I., Beasley, R., Don Gelosh, E., Mimi Heisey Ian Presland, C.-A., & Lori Zipes, E. (2018). INCOSE SYSTEMS ENGINEERING COMPETENCY FRAMEWORK Compiled and Edited by: Primary Contributing Authors (alphabetical order) (tech. rep.).
- Price, H. E. (1985). The Allocation of Functions in Systems (tech. rep. No. 1).
- Prins, M., Koolwijk, J., Volker, L., Wamelink, H., Prins, i. M., Koolwijk, I. J., Volker, I. L., & ir JWF Wamelink, d. (2014). *Briefing: Static or dynamic?* (Tech. rep.). https://www.researchgate.net/publication/234022330
- PwC. (2019). The road to circularity: why a circular economy is becoming the new normal. Price Waterhouse Cooper: London, UK. https://www.pwc.nl/en/assets/ documents/pwc-the-road-to-circularity-en.pdf
- Qazi, A., Quigley, J., Dickson, A., & Kirytopoulos, K. (2016). Project Complexity and Risk Management (ProCRiM): Towards modelling project complexity driven risk paths in construction projects. *International Journal of Project Management*, 34(7), 1183–1198. https://doi.org/10.1016/J.IJPROMAN.2016.05.008
- Rahla, K. M., Bragança, L., & Mateus, R. (2019). Obstacles and barriers for measuring building's circularity. *IOP Conference Series: Earth and Environmental Science*, 225(1). https://doi.org/10.1088/1755-1315/225/1/012058

- Rahla, K. M., Mateus, R., & Bragança, L. (2021). Selection Criteria for Building Materials and Components in Line with the Circular Economy Principles in the Built Environment—A Review of Current Trends. *Infrastructures 2021, Vol. 6, Page 49, 6*(4), 49. https://doi.org/10.3390/INFRASTRUCTURES6040049
- Reid, J., & Wood, D. (2023). Systems engineering applied to urban planning and development: A review and research agenda. Systems Engineering, 26(1), 88–103. https://doi.org/10.1002/SYS.21642
- RIBAJ. (2022). How to test and validate your sustainable design decisions RIBAJ. https://www.ribaj.com/intelligence/designing-for-the-climate-emergency-testing-and-validating-your-design-decisions
- Rose, T. M., & Manley, K. (2014). Revisiting the adoption of innovative products on Australian road infrastructure projects. Construction Management and Economics, 32(9), 904–917. https://doi.org/10.1080/01446193.2014.938670
- Sanchez, B., & Haas, C. (2018). Capital project planning for a circular economy. Construction Management and Economics, 36(6), 303–312. https://doi.org/10.1080/ 01446193.2018.1435895
- Schützenhofer, S., Kovacic, I., Rechberger, H., & Mack, S. (2022). Improvement of Environmental Sustainability and Circular Economy through Construction Waste Management for Material Reuse. Sustainability (Switzerland), 14(17). https://doi.org/10.3390/SU141711087
- Senaratne, S., Abhishek, K. C., Perera, S., & Almeida, L. (2021). Promoting stakeholder collaboration in adopting circular economy principles for sustainable construction. World Construction Symposium, 471–482. https://doi.org/10.31705/WCS.2021.41
- Shen, W., Hao, Q., Mak, H., Neelamkavil, J., Xie, H., Dickinson, J., Thomas, R., Pardasani, A., & Xue, H. (2010). Systems integration and collaboration in architecture, engineering, construction, and facilities management: A review. Advanced Engineering Informatics, 24(2), 196–207. https://doi.org/10.1016/J.AEI.2009.09. 001
- Shooshtarian, S., Maqsood, T., Wong, P., Caldera, S., Ryley, T., Zaman, A., & Cáceres Ruiz, A. (2024). Circular economy in action: the application of products with recycled content in construction projects – a multiple case study approach. Smart and Sustainable Built Environment, 13(2), 370–394. https://doi.org/10.1108/ SASBE-08-2023-0213
- Snyder, H. (2019). Literature review as a research methodology: An overview and guidelines. Journal of Business Research, 104, 333–339. https://doi.org/10.1016/J. JBUSRES.2019.07.039
- Sparrevik, M., de Boer, L., Michelsen, O., Skaar, C., Knudson, H., & Fet, A. M. (2021). Circular economy in the construction sector: advancing environmental performance through systemic and holistic thinking. *Environment Systems and Deci*sions, 41(3), 392–400. https://doi.org/10.1007/s10669-021-09803-5
- Tingley, D. D. (2014). Reducing Material Demand in Construction. www.withbotheyesopen. com,
- Többen, J., & Opdenakker, R. (2022). Developing a Framework to Integrate Circularity into Construction Projects. Sustainability (Switzerland), 14(9). https://doi.org/ 10.3390/su14095136
- Tookey, J. E., Kulatunga, K., Kulatunga, U., Amaratunga, D., & Haigh, R. (2011). Client's championing characteristics that promote construction innovation. *Con*-

struction Innovation, 11(4), 380–398. https://doi.org/10.1108/14714171111175873/ FULL/PDF

- Treggiden, K. (2020). StoneCycling Turns Industrial and Demolition Waste Into Bricks. https://design-milk.com/circular-by-design-stonecycling-turns-industrial-and-demolition-waste-into-bricks/
- UKGBC. (2019). Circular economy guidance for construction clients (tech. rep.). (UK Green Building Council).
- US Department of Defense. (2001). Systems Engineering Fundamentals Supplementary Text Prepared By The Defense Acquisition (tech. rep.). UNIVERSITY PRESS FORT BELVOIR, VIRGINIA 22060-5565.
- van Stijn, A., & Gruis, V. (2020). Towards a circular built environment: An integral design tool for circular building components. Smart and Sustainable Built Environment, 9(4), 635–653. https://doi.org/10.1108/SASBE-05-2019-0063/FULL/PDF
- Van Den Berg, M., Voordijk, H., & Adriaanse, A. (2021). BIM uses for deconstruction: an activity-theoretical perspective on reorganising end-of-life practices. *Construction Management and Economics*, 39(4), 323–339. https://doi.org/10.1080/01446193. 2021.1876894
- Van Der Meer, J., Hartmann, A., Van Der Horst, A., & Dewulf, G. (2015). Challenges of using systems engineering for design decisions in large infrastructure tenders. *Engineering Project Organization Journal*, 5(4), 133–145. https://doi.org/10. 1080/21573727.2015.1113401
- van der Zwaag, M., Wang, T., Bakker, H., van Nederveen, S., Schuurman, A. C., & Bosma, D. (2023). Evaluating building circularity in the early design phase. Automation in Construction, 152. https://doi.org/10.1016/j.autcon.2023.104941
- Vandenbroucke, M. L. (2016). Design, dimensioning and evaluation of demountable building elements. https://researchportal.vub.be/en/publications/design-dimensioningand-evaluation-of-demountable-building-elemen
- Vandervaeren, C., Galle, W., Stephan, A., & De Temmerman, N. (2022). More than the sum of its parts: Considering interdependencies in the life cycle material flow and environmental assessment of demountable buildings. *Resources, Conservation and Recycling*, 177. https://doi.org/10.1016/j.resconrec.2021.106001
- Vehring, S. (2023). Role of Conformity Assessment in the Circular Economy Outlook IAF. https://iaf.news/2023/08/30/role-of-conformity-assessment-in-the-circulareconomy/
- Watson, M., Mesmer, B., Roedler, G., Rousseau, D., Calvo-Amodio, J., Keating, C., Miller, W. D., Lucero, S., Gold, R., Jones, C., Long, D., Russell, R. W., & Sedmak, A. (2022). Systems Engineering Principles (tech. rep.). https://www.incose.org/ publications/products/se-principles
- Watson, M. D., Mesmer, B., & Farrington, P. (2016). Engineering Elegant Systems: Postulates, Principles, and Hypotheses of Systems Engineering (tech. rep.).
- Wheeler, W. (2004). System Requirements and Project Success. Integrating Wireless Technology in the Enterprise, 125–134. https://doi.org/10.1016/B978-155558295-1/50008-X
- Wood, H., & Ashton, P. (2010). The factors of project complexity, 69–80. https://research.brighton.ac.uk/en/publications/the-factors-of-project-complexity
- World Architects. (2017). MycoTree KIT Karlsruhe + ETH Zürich + Singapore-ETH Centre. https://www.world-architects.com/en/architecture-news/works/ mycotree

- World Economic Forum. (2020). Nature must be the most important stakeholder this decade World Economic Forum. https://www.weforum.org/agenda/2020/01/why-nature-will-be-the-most-important-stakeholder-in-the-coming-decade/
- Yeheyis, M., Hewage, K., Alam, M. S., Eskicioglu, C., & Sadiq, R. (2013). An overview of construction and demolition waste management in Canada: A lifecycle analysis approach to sustainability. https://doi.org/10.1007/s10098-012-0481-6
- Zhu, W., Fan, X., & He, Q. (2020). A hierarchical and process-oriented framework for disassemblability evaluation in product maintainability design. International Journal of Advanced Manufacturing Technology, 110(3-4), 777–795. https://doi.org/10. 1007/S00170-020-05830-9/TABLES/6